An obscured quasar census with the 4MOST IR AGN survey: design, predicted properties, and scientific goals

Carolina Andonie⁽¹⁾,^{1,2}* David M. Alexander,¹ Claire Greenwell⁽¹⁾,¹ Sotiria Fotopoulou⁽²⁾,³ Ryan Hickox,⁴ David J Rosario,⁵ Carolin Villforth,⁶ Johannes Buchner,² Jens-Kristian Krogager⁽²⁾,^{7,8} Brivael Laloux,^{2,9} Andrea Merloni,² Mara Salvato⁽²⁾,² Ole Streicher¹⁰ and Wei Yan¹¹

¹Centre for Extragalactic Astronomy, Department of Physics, Durham University, Durham DH1 3LE, UK

²Max-Planck-Institut für Extraterrestrische Physik, Gießenbachstraße, D-85748 Garching, Germany

³School of Physics, HH Wills Physics Laboratory, University of Bristol, Tyndall Avenue, Bristol BS8 1TL, UK

⁵School of Mathematics, Statistics and Physics, Newcastle University, Newcastle upon Tyne, NE1 7RU, UK

- ⁸Centre de Recherche Astrophysique de Lyon, Université de Lyon 1, UMR5574, 9 Av Charles André, 69230 Saint-Genis-Laval, France
- ⁹INAF Osservatorio Astronomico di Capodimonte, Via Moiariello 16, I-80131 Napoli, Italy

¹⁰Leibniz Institute for Astrophysics Potsdam (AIP), An der Sternwarte 16, D-14482 Potsdam, Germany

¹¹Wartburg College, Science Center, 100 Wartburg Blvd. Waverly, IA 50677, USA

Accepted 2025 April 10. Received 2025 April 4; in original form 2024 September 30

ABSTRACT

We present the 4MOST (4-metre Multi-Object Spectroscopic Telescope) infrared (IR) AGN survey, the first large-scale optical spectroscopic survey characterizing mid-infrared (MIR) selected obscured active galactic nuclei (AGNs). The survey targets $\approx 212\,000$ obscured IR AGN candidates over $\approx 10\,000 \, \text{deg}^2$ down to a magnitude limit of $r_{AB} = 22.8$ mag and will be $\approx 100 \times$ larger than any existing obscured IR AGN spectroscopic sample. We select the targets using an MIR colour criterion applied to the unWISE catalogue from the *WISE* (*Wide-field Infrared Survey Explorer*) all-sky survey, and then apply a $r - W2 \ge 5.9$ mag cut; we demonstrate that this selection will mostly identify sources obscured by $N_{\rm H} > 10^{22} \, \text{cm}^{-2}$. The survey complements the 4MOST X-ray survey, which will follow up ~ 1 M *eROSITA* (extended ROentgen Survey with an Imaging Telescope Array)-selected (typically unobscured) AGN. We perform simulations to predict the quality of the spectra that we will obtain and validate our MIR–optical colour-selection method using X-ray spectral constraints and UV-to-far-IR spectral energy distribution (SED) modelling in four well-observed deep-sky fields. We find that: (1) $\approx 80 - 87$ per cent of the *WISE*-selected targets are AGN down to $r_{AB} = 22.1 - 22.8$ mag of which ≈ 70 per cent are obscured by $N_{\rm H} > 10^{22} \, \text{cm}^{-2}$, and (2) ≈ 80 per cent of the 4MOST IR AGN sample will remain undetected by the deepest *eROSITA* observations due to extreme absorption. Our SED-fitting results show that the 4MOST IR AGN survey will primarily identify obscured AGN and quasars (≈ 55 per cent of the sample is expected to have $L_{AGN,IR} > 10^{45} \, \text{erg s}^{-1}$) residing in massive galaxies ($M_{\star} \approx 10^{10} - 10^{12} \, \text{M}_{\odot}$) at $z \approx 0.5 - 3.5$ with ≈ 33 per cent expected to be hosted by starburst galaxies.

Key words: surveys-quasars: general-infrared: galaxies.

1 INTRODUCTION

Supermassive black holes (SMBHs) are found at the centres of most massive galaxies (e.g. Magorrian et al. 1998; Marconi & Hunt 2003), and are thought to play an important role in galaxy evolution (e.g. Alexander & Hickox 2012; Kormendy & Ho 2013). Most SMBH growth occurs in a phase of efficient accretion, where SMBHs are observed as active galactic nuclei (AGNs), or in extremely luminous cases, as quasars.¹ The tight relationships between the SMBH mass

¹In this work, we refer to quasars as AGNs with a 8 – 1000 μ m (IR) AGN luminosity $L_{AGN,IR} > 10^{45}$ erg s⁻¹, which is approximately equivalent to

and galaxy properties such as the stellar velocity dispersion and the mass of the bulge provide compelling observational evidence supporting a connection between SMBH accretion and the evolution of their host galaxies (e.g. Magorrian et al. 1998; Ferrarese & Merritt 2000; Gebhardt et al. 2000; Marconi & Hunt 2003; Kormendy & Ho 2013; Madau & Dickinson 2014). The key early growth phase of galaxies and their SMBHs is expected to be predominantly obscured (Hopkins et al. 2008; Alexander & Hickox 2012; Blecha et al. 2018; Hickox & Alexander 2018), with large reservoirs of gas

an AGN bolometric luminosity of $L_{AGN,bol} > 3 \times 10^{45}$ erg s⁻¹, regardless of whether they are optically bright and unobscured or optically faint and obscured.

© 2025 The Author(s).

Published by Oxford University Press on behalf of Royal Astronomical Society. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

⁴Department of Physics and Astronomy, Dartmouth College, 6127 Wilder Laboratory, Hanover, NH 03755, USA

⁶Department of Physics, University of Bath, Claverton Down, Bath BA2 7AY, UK

⁷ French-Chilean Laboratory for Astronomy, 7591245 Santiago, IRL 3386, CNRS and U. de Chile, Casilla 36-D, Chile

^{*} E-mail: cpandonie@mpe.mpg.de

and dust feeding both the SMBH and the galaxy. Hence, obscured quasars provide key information on the parallel evolution between SMBHs and galaxies. However, due to observational limitations, the reliable identification and characterization of obscured quasars is challenging, presenting a significant hurdle when investigating SMBH and galaxy growth.

So far, the majority of the sizeable, large-scale AGN optical spectroscopic surveys target AGN and guasars that are luminous in the optical band (e.g. Rakshit, Stalin & Kotilainen 2020; Alexander et al. 2023; Storey-Fisher et al. 2024), such as surveys from the Sloan Digital Sky Survey (SDSS, York et al. 2000) and the Dark Energy Spectroscopic Instrument (DESI, DESI Collaboration 2016). SDSS guasar surveys have made a remarkable contribution to our AGN knowledge characterizing accretion disc (AD) physics (e.g. LaMassa et al. 2015; Lusso & Risaliti 2016), the broad emission-line region (e.g. Nagao, Marconi & Maiolino 2006; Temple et al. 2021, 2023), the broad absorption-line region (e.g. Trump et al. 2006; Rankine et al. 2020; Petley et al. 2022), the clustering properties of quasars (e.g. Laurent et al. 2017; Powell et al. 2020), and optically bright Type 2 AGN (e.g. Zakamska et al. 2003, 2004, 2008), among other key results. However, due to sensitivity limitations, SDSS only targets optically bright AGN; hence, it misses the wide majority of the faint, obscured systems, as the optical band is strongly affected by absorption. This presents a particular challenge for understanding the full AGN population as it is now known that the majority of the AGN population is obscured (e.g. Martinez-Sansigre et al. 2006; Fiore et al. 2008; Ueda et al. 2014; Aird et al. 2015; Ricci et al. 2015; Laloux et al. 2024).

X-ray observations are arguably one of the most efficient methods to identify AGN, including the obscured AGN population, for several reasons (see Brandt & Alexander 2015): (1) X-ray emission appears to be a near-universal property of AGN, (2) X-ray photons can penetrate relatively high column densities of gas with $N_{\rm H} \approx$ $10^{21} - 10^{24}$ cm⁻², identifying obscured AGN that optical selections would miss, and (3) the contamination from the host galaxy emission in the X-ray band is negligible in most cases. Furthermore, the Xray emission from the entire AGN population is imprinted on the cosmic X-ray background (e.g., Comastri et al. 1995; Gilli, Comastri & Hasinger 2007; Ueda et al. 2014; Ananna et al. 2019), providing a key probe of SMBH accretion in the universe. In the local universe, we have an excellent census of the obscured AGN population with the BAT AGN Spectroscopic Survey (BASS, Koss et al. 2017), based on the 70- and 105-months Swift-Burst Alert Telescope (BAT) All-sky Hard X-Ray survey (Baumgartner et al. 2013; Oh et al. 2018). BASS is an all-sky survey in the ultra-hard X-ray band (14 - 195 keV) able to penetrate very large levels of absorption, even in the Comptonthick (CT, $N_{\rm H} > 10^{24} \,{\rm cm}^{-2}$) regime (e.g. Ricci et al. 2015). BASS provides optical spectroscopy for an almost obscuration-unbiased AGN sample at least up to $N_{\rm H} \approx 10^{24} \, {\rm cm}^{-2}$ in the local universe (z < 0.2) or for relatively bright X-ray sources (14–195 keV limiting depth of $8.40 \times 10^{-12} \text{ erg s}^{-1} \text{ cm}^{-2}$, Oh et al. 2018). However, at larger redshifts, due to sensitivity limitations, when the level of obscuration is CT, even the deepest X-ray surveys will miss most heavily obscured and CT AGN (e.g. Treister et al. 2004; Alexander et al. 2008; Lansbury et al. 2017).

An alternative method for identifying obscured AGN involves using the mid-infrared (MIR) band. Bright MIR emission is a key signature of SMBH accretion, and arises from the reprocessing of optical/UV radiation produced within the AD by circumnuclear dust (the 'dusty torus'). Hence, MIR emission is not strongly affected by absorption and is able to identify obscured AGN that may escape optical and even X-ray selection approaches (e.g. Donley et al. 2012; Del Moro et al. 2016; Andonie et al. 2022). However, a challenge with the MIR band is that it can be highly contaminated by star formation (SF) emission from the host galaxy, making AGN identification challenging, in particular when the host galaxy dominates the MIR emission. To address this problem, a number of MIR AGN identification techniques have been developed. One of the most popular criteria is based on MIR colours, which are sensitive to the dust temperature and, hence, can be successfully used to distinguish AGN from star-forming galaxies (e.g. Stern et al. 2005; Lacy et al. 2007; Donley et al. 2012; Mateos et al. 2012; Assef et al. 2018). Currently, AGN selection approaches using MIR colours are typically defined using photometry from the Spitzer (Werner et al. 2004) and Wide-field Infrared Survey Explorer (WISE, Wright et al. 2010) telescopes. WISE has made a significant contribution to the discovery of luminous MIR quasars, which are typically characterized by red MIR colours. There are various AGN selection techniques based on W1 and W2 WISE magnitudes that reliably identify AGN (e.g. Stern et al. 2012; Assef et al. 2018). Given that WISE is an all-sky survey, these selections have the potential to find a very large number of AGN across the entire sky.

Multiple studies have published spectroscopic catalogues based on MIR-selected AGN. For example, there are several dedicated spectroscopic surveys following up MIR-selected AGN in small fields of the sky, containing several hundred up to a few thousand objects (e.g. Lacy et al. 2013; Kochanek et al. 2012; Li et al. 2024; Wang et al. 2025). A few works have also studied samples of a few thousand *WISE*-selected AGN across the entire sky, using archival optical spectroscopic data (e.g. DiPompeo et al. 2018; Carroll et al. 2023); hence, those samples do not provide a uniform and homogeneous spectroscopically identified MIR AGN sample. Consequently, large-scale spectroscopic surveys targeting MIR-selected AGN are, so far, non-existent.

Recently, several studies have explored the properties of MIRselected AGNs finding interesting results that, as a whole, provide strong evidence for an evolutionary connection between SMBHs and galaxies. First, obscured and unobscured quasars seem to reside in different large-scale environments, where (on average) obscured sources are hosted in more massive haloes and are more clustered than their unobscured counterparts (e.g. Hickox et al. 2011; DiPompeo et al. 2014; Powell et al. 2018; Petter et al. 2023). Second, the host galaxies of obscured quasars have larger star formation rates (SFRs) than unobscured quasars (e.g. Chen et al. 2015; Andonie et al. 2022), and are more likely to reside in compact starbursts which can produce an obscuration comparable with that of the dusty torus (e.g. Circosta et al. 2019; Andonie et al. 2024). Third, IR AGN are more likely to be found in close galaxy pairs compared to optically or X-rayselected systems which are more typically unobscured AGN (e.g. Satyapal et al. 2014; Dougherty et al. 2024; La Marca et al. 2024). These results indicate environmental differences between obscured and unobscured AGN, suggesting that at least some fraction of the obscured MIR AGN population represents a distinct phase in the overall evolution of galaxies, a hypothesis proposed by several observational and theoretical studies (e.g. Sanders et al. 1988; Di Matteo, Springel & Hernquist 2005; Hopkins et al. 2008). However, this scenario has been difficult to directly test since (1) many of the aforementioned studies have been performed in deep but small fields (and hence they are limited in source statistics), or (2) they lack spectroscopic confirmation, relying on more uncertain photometric redshifts.

Here, we present the 4MOST² IR AGN survey (PI: D. M. Alexander), the first large-scale optical spectroscopic survey characterizing the MIR-selected obscured AGN population. The objective of the 4MOST IR AGN survey is to provide the first large and complete obscured MIR AGN census up to redshift ~ 1 with a spectroscopically identified IR AGN sample $\sim 100 \times$ larger than existing studies. This will allow for the most statistically robust characterization of the obscured IR AGN population, including identifying new AGN subpopulations and testing different AGN-host galaxy evolutionary scenarios. The 4MOST IR AGN survey is designed to complement the 4MOST X-ray AGN survey (PI: A. Merloni, Merloni et al. 2019), which targets AGNs selected by eROSITA (extended ROentgen Survey with an Imaging Telescope Array; Merloni et al. 2012; Predehl et al. 2021). eROSITA has a predominantly soft X-ray (0.5-2 keV) response, and largely identifies luminous unobscured AGN (Liu et al. 2022), while the 4MOST IR AGN survey will predominantly identify the obscured AGN missed by eROSITA. Aside from the 4MOST IR AGN survey, there are other surveys that have the potential to investigate obscured AGNs such as DESI (Alexander et al. 2023; Fawcett et al. 2023), the Optical, Radio Continuum and HI Deep Spectroscopic Survey (ORCHIDSS; Duncan et al. 2023), the 4MOST Gaia Purely Astrometric Quasar Survey (PAQS; Krogager et al. 2023), and the Chilean AGN/Galaxy Extragalactic Survey (Bauer et al. 2023). All these surveys combined will open up new grounds for the exploration of the optical spectroscopic properties of obscured IR AGNs. For more information on the 4MOST capabilities and instrumental specifications, see de Jong et al. (2019).

This paper presents the design of the 4MOST IR AGN survey, our target selection approach, and predictions for the AGN and host galaxy properties of the identified sources. The structure of the paper is as follows. In Section 2, we describe the datasets used to construct our catalogue and to test our AGN selection. In Section 3, we describe the 4MOST IR AGN survey design, describing the obscured IR AGN selection function and the methods used to test their performance. In Section 4, we then present our final catalogue, the predictions for the properties of the survey, and the scientific goals of our survey. Finally, Section 5 summarizes this work.

Throughout this work, we adopt a concordance cosmology (Hinshaw et al. 2013), and a Chabrier initial mass function (Chabrier 2003). The magnitudes are in Vega units unless we state otherwise.

2 DATA SETS AND METHODS USED TO DEFINE THE 4MOST IR AGN SURVEY

This section describes the different data sets used to define the 4MOST IR AGN survey. In brief, to identify obscured IR AGN candidates, we first select AGN using the Assef et al. (2018) 90 per cent AGN reliability colour selection technique (hereafter A18 R90), which requires a 3σ detection in W1 and a 5σ detection in W2, and selects the sources that meet the following criteria

$$W1 - W2 > \begin{cases} \alpha_R \exp\{\beta_R (W2 - \gamma_R)^2\}, \ W2 > \gamma_R, \\ \alpha_R, \ W2 \le \gamma_R, \end{cases}$$
(1)

²4MOST (4-metre Multi-Object Spectroscopic Telescope; de Jong et al. 2019) is a wide-field spectroscopic survey facility which will be mounted at the VISTA telescope, and is expected to obtain optical spectra for more than 80 million sources over an area of > 17000 deg⁻² with a 5-yr survey starting late 2025/early-2026. 4MOST will complement large multi-wavelength surveys such as those undertaken by *eROSITA*, *Gaia*, and *Euclid*, as well as groundbased optical imaging and spectroscopic facilities such as LSST, SDSS, and DESI. where $\alpha_R = 0.65$, $\beta_R = 0.153$, and $\gamma_R = 13.86$ (see also Fig. A1 of Appendix A). We then select the obscured sources using an optical-to-MIR colour cut, following Hickox et al. (2007). While the entire survey covers ~ 10 000 deg² of the sky, we test and optimize the selection in small but well-observed fields using UV-to-FIR spectral energy distribution (SED) template fitting and X-ray spectral constraints. Sections 2.1 and 2.2 describe the optical and IR data sets used to perform the target selection. Sections 2.3 and 2.4 describe the data sets and methods used to test the AGN reliability of our selection approach, and predict the AGN and host galaxy properties of the sample.

2.1 Parent sample: DESI Legacy Surveys DR10

The parent sample of the 4MOST IR AGN survey is the Legacy Surveys (LS) DR10 South.³ The original LS map 14000 deg² of the sky in the optical bands g, r, and z (Dev et al. 2019). LS is primarily constructed using three imaging surveys. The observations in the South Galactic Cap are conducted by The Dark Energy Camera Legacy Survey and are carried out by the Dark Energy Camera (DECam; Flaugher et al. 2015) at the Blanco Telescope at the Cerro Tololo Inter-American Observatory. In the North Galactic Cap, the observations are conducted by the Beijing-Arizona Sky Survey (BASS) and the Mayall z-band Legacy Survey (MzLS). The BASS observations were taken at the University of Arizona Steward Observatory 2.3 m (90 inch) Bart Bok Telescope at the Kitt Peak National Observatory, and the MzLS observations are taken at the Mayall Telescope at the Kitt Peak National Observatory. LS DR10 expanded the original LS to $> 20000 \text{ deg}^2$, adding optical *i*-band DECam observations from NOIRLab in the Southern Galactic Cap. LS DR10 also provides deblended WISE photometry in the W1, W2, W3, and W4 bands, which is point spread function (PSF)-fitted and measured at the position of the griz-detected sources (Dey et al. 2019). Overall, the LS DR10 contains \sim 2.8 billion unique sources.

The depth of the LS is not uniform and depends on the number of exposures in the area. The point-source 5σ depths for two observations in AB magnitudes⁴ (90 per cent of the survey has at least two observations in *grz*; Dey et al. 2019) are g = 24.7, r = 23.9, i = 23.5, and z = 23.

2.2 WISE data and matching with LS DR10 catalogue

As previously mentioned, we identify the IR AGN sample using the A18 R90 selection, which was calibrated using AllWISE photometry. Since the publication of A18, newer *WISE* catalogues provide deeper *WISE* photometry and larger numbers of *WISE* sources, which could potentially provide more complete AGN samples. Hence, to optimize our IR AGN selection, we test the efficiency of the A18 R90 selection for three different *WISE* datasets: the AllWISE⁵ catalogue, the unWISE catalogue (Schlafly, Meisner & Green 2019), and the LS DR10-derived *WISE* catalogue. After performing extensive tests, we decided to adopt the unWISE fluxes to identify the IR AGN in the 4MOST IR AGN survey. See Appendix A for our tests and justifications of this choice, in addition to the details of each catalogue and their efficiency in identifying AGN. Since the LS DR10 is our parent sample and provides the optical positions of the 4MOST IR AGN survey, we need to find the unWISE

³See the LS DR10 description here: https://www.legacysurvey.org/dr10/ description/.

⁴See https://www.legacysurvey.org/dr10/description/#depths for more details ⁵http://wise2.ipac.caltech.edu/docs/release/AllWISE/

Field	Photometric catalogue	UV-to-NIR photometry	X-ray catalogue
COSMOS	COSMOS2015 (Laigle et al. 2016) and Jin et al. (2018) IR catalogue	- CFHT: <i>u</i> * - Subaru Suprime-Cam: <i>V</i> , <i>B</i> , <i>r</i> +, <i>i</i> +, <i>z</i> + + - UltraVISTA-DR2: <i>Y</i> , <i>J</i> , <i>H</i> , and <i>Ks</i>	Chandra X-ray spectral constraints from Laloux et al. (2023)
XMM-LSS	HELP (Shirley et al. 2021)	 CFHT/Megacam:U Subaru/Suprime: g, r, i, z VISTA: Y, J, H, and Ks 	<i>XMM–Newton</i> point-source catalogue from Chiappetti et al. (2018)
Boötes	HELP (Shirley et al. 2021)	 Large Binocular Telescope Observatory: U KPNO/Mosaic: B, r, i, z NOAO/NEWFIRM: J, H, and K 	<i>Chandra</i> point-source catalogue from Masini et al. (2020)
ELAIS-S1	HELP (Shirley et al. 2021)	- CTIO/DECam: g, r, i, z - VISTA: Y, J, H , and Ks .	<i>XMM–Newton</i> point source catalogue from Ni et al. (2021) and spectral constraints from Yan et al. (2023)

Table 1. Multiwavelength data sets used in each field to perform the UV-to-FIR SED-fitting and the X-ray analyses.

counterparts of the LS sources. The full width at half-maximum of the W1 and W2 PSFs are $\approx 6 \operatorname{arcsec}$ (Wright et al. 2010) with typical positional uncertainties ranging $0.4-1 \operatorname{arcsec}^{6}$ while the LS optical PSF is typically $\sim 1 \operatorname{arcsec}$, with a typical positional uncertainty of 0.1 arcsec for AGN-like sources (e.g. Salvato et al. 2022). Hence, we must be careful when matching the catalogues to avoid source misidentification. To do this, we find all the matches between the WISE and the optical LS positions within 2 arcsec separation. When one WISE source has multiple optical counterparts, we pick the optical source with the largest W2 flux in the LS DR10 catalogue, which is associated with the central galaxy instead of an extended substructure. We chose a 2 arcsec matching radius after visualizing the unWISE and LS images of a random sample of our sources. We find that the typical distance between the MIR and optical positions for the same source is always $< 2 \operatorname{arcsec}$; hence, a larger matching radius would significantly increase the number of source misidentifications between two point-like sources. In addition, we calculate the probability of having a spurious match within a 2 arcsec matching radius based on the source density of A18-selected AGNs ($\sim 140 \text{ deg}^{-2}$; see Table A1 in Appendix A), and we find that it is only ~ 0.05 per cent. Furthermore, we find that the probability of a true match between an optical and WISE position at >2 arcsec separation, given the WISE and LS positional uncertainties, is < 3 per cent, which is consistent with that found in previous works (e.g. Greenwell et al. 2024). A 2 arcsec matching radius between WISE and optical catalogues has also been found by previous work to be a good compromise to minimize the spurious matches (e.g. Mateos et al. 2012; Assef et al. 2013, 2018; Greenwell et al. 2024). We apply the same method to find the LS counterparts of AllWISE sources in our comparisons.

2.3 UV-to-FIR SED fitting

To test the AGN reliability of our obscured IR AGN colour selection approach and also predict the properties of the 4MOST IR AGN survey, we perform detailed UV-to-FIR SED template fitting in four deep and well-observed fields in the sky: the Cosmic Evolution Survey (COSMOS) field, the XMM-Newton Large-Scale Structure (XMM-LSS) field, the Boötes field, and the European Large Area Infrared Survey field South 1(ELAIS-S1); see Table 1. These fields were selected because they have UV-to-far-IR (FIR) photometric coverage and X-ray spectral observations. This subsection describes the photometric data sets and the SED modelling adopted in this work.

2.3.1 Multiwavelength photometric data sets

The photometry used to perform SED fitting was taken from the Herschel Extragalactic Legacy Project (HELP, Shirley et al. 2019, 2021) for the Boötes, XMM-LSS, and ELAIS-S1 fields. The HELP survey provides homogeneous optical, near-IR (NIR), MIR, and FIR photometry for the *Herschel* Spectral and Photometric Imaging Receiver (SPIRE) extragalactic survey fields: the Herschel Multitiered Extragalactic Survey (Oliver et al. 2012) and the Herschel Atlas survey (Eales et al. 2010). In the case of COSMOS, we take the UV-to-NIR photometry from the COSMOS2015 catalogue (Laigle et al. 2016), and the MIR and FIR photometry from the Jin et al. (2018) catalogue. For all sources, we take the spectroscopic redshift when available, otherwise, we use the photometric redshift (see Shirley et al. 2021 for details on the redshifts of the HELP fields, and section 2.1 of Andonie et al. 2022 for details on the redshift assignment for the COSMOS field).

The available UV-to-NIR photometry varies for each field since they have been observed by different instruments, but the MIR and FIR observations come from the same telescopes: *Spitzer* and *Herschel*. The MIR and FIR photometry includes the fluxes from the following bands: *Spitzer* Infrared Array Camera (IRAC) fluxes at 3.6, 4.5, 5.8, and 8 μ m; *Spitzer* Multiband Imaging Photometer (MIPS) at 24 μ m; *Herschel* Photodetector Array Camera and Spectrometer (PACS) at 100 and 160 μ m; and *Herschel* SPIRE at 250, 350, and 500 μ m. Table 1 summarizes the different UV-to-NIR data sets used in each field.

2.3.2 SED modelling

To perform SED fitting, we use the multicomponent Bayesian SEDfitting code FORTESFIT⁷ (Rosario 2019). FORTESFIT is a publicly available PYTHON-based hierarchical Bayesian modelling framework for multi-wavelength SED fitting. It allows the incorporation of userdefined SED models and the combination of these in a fit with flexible specifications of the prior distribution on modelling parameters. FORTESFIT also allows the construction of full posterior distributions of all free parameters, providing accurate parameter uncertainties. It

⁷https://github.com/vikalibrate/FortesFit

⁶The *WISE* positional uncertainties depend on the S/N of the sources. Before matching the catalogues, we pre-select the *WISE* sources with *W*1 S/N>3 and *W*2 S/N>5 (see Section 3). We find that 95 per cent of the *WISE* sources have a *W*1 S/N>20. Hence, we expect the positional uncertainties of our sources to be typically < 0.5 arcsec. See the WISE All-Sky Data Release Explanatory Supplement for details (https://wise2.ipac.caltech.edu/ docs/release/allsky/expsup/sec6_4.html).

Component	Parameter	Description	Range	Reference
Stellar population emission (SP)	$\log M_{\star} [\mathrm{M}_{\odot}]$	Stellar mass	[7, 13]	Bruzual & Charlot (2003)
	$\log t'$ [Gyr]	Age of the stellar population	[8, 10.1]	
	$\log \tau [\log \text{Gyr}]$	Star formation time-scale	[0.01, 15]	
	$E(B-V)_{\rm SP}$ [mag]	Galaxy reddening	[0, 0.5]	Calzetti et al. (2000)
Accretion disc (AD)	$\log L_{2500} [\text{erg s}^{-1} \text{Hz}^{-1}]$	Luminosity at 2500 Å	[26,36]	Richards et al. (2006)
	$E(B-V)_{\rm AD}$ [mag]	AD reddening	[0, 1]	Prevot et al. (1984)
Torus	$\log L_{AGN,IR} [erg s^{-1}]$	$8 - 1000 \mu m$ AGN luminosity	[38,48]	Mullaney et al. (2011)
	Γ_{s}	Short-wavelength slope	[-0.3, 1.4]	• • •
	$\Gamma_{\rm L}$	Long-wavelength slope	[-1, 0.5]	
	$\lambda_{\text{Brk}} \left[\mu m \right]$	Turnover wavelength	17.5*	
	$\lambda_{BB} [\mu m]$	Blackbody peak wavelength	40*	
	$T_{\rm BB,s}$ [K]	Hot dust temperature	1500*	
Star formation (SF)	$\log L_{\rm SF,IR}$ [erg s ⁻¹]	$8 - 1000 \mu m$ star formation luminosity	[42,48]	Dale et al. (2014)
	$\alpha_{\rm SF}$	Shape parameter	[0.06,4]	

Table 2. Description and allowed ranges of the parameters of each component identified in our SED-fitting models. For more details on each model, please refer to the references. Parameters with a * symbol are fixed.

has been successfully used in several works to measure the physical properties of AGNs and galaxies from fitting their broad-band UV-to-radio SEDs (e.g. Rosario 2019; Scholtz et al. 2021; Andonie et al. 2022, 2024; Laloux et al. 2023).

In our implementation of the SED fitting for our sample, we use FORTESFIT along with a set of models that feature emission from the AD and torus of the AGN, as well as stellar and dust components of the host galaxy (see below). FORTESFIT does not explicitly assume energy balance in the galaxy emission, as performed in some other SED-fitting codes (e.g. CIGALE). As accurate galaxy star formation histories are not the main focus of this analysis, strict energy balance is not a critical requirement for our fits.

In our FORTESFIT implementation, we use the PYTHON adaptation (Buchner et al. 2014) of the Nested Sampling algorithm MULTINEST (Feroz & Hobson 2008; Feroz, Hobson & Bridges 2009; Feroz et al. 2019) as a fitting engine. FORTESFIT uses the likelihood function outlined by Sawicki (2012), which accounts for photometric detections and upper limits self-consistently by treating them as a combination of independent Gaussian likelihoods (i.e. a blend of χ^2 and error functions).

We adopt a similar SED-fitting approach to that taken in Andonie et al. (2022) to identify the AGN and to model the AGN and host galaxy properties of our sample. Here, we fit the observed-frame $0.3 - 500 \ \mu m$ SED of the objects. In modelling the SEDs, we include the four following components: (1) the unabsorbed stellar emission characterized by the Bruzual & Charlot (2003) stellar-population models, attenuated according to the Milky Way reddening law from Calzetti et al. (2000); (2) the UV emission from the AD following the empirical model of Richards et al. (2006) and the Pei (1992) reddening law for the Small Magellanic Clouds; (3) the IR emission from the AGN torus, based on the empirical DECOMPIR AGN model of Mullaney et al. (2011); and (4) a semi-empirical star-forming galaxy model which reproduces the full range of dust temperatures observed in galaxies (Dale et al. 2014).

To break the degeneracy between fitting the AD and the stellar emission in the rest-frame optical–NIR waveband, in the case of X-ray-detected sources, we use the well-established $L_{2500}-L_{2keV}$ relation for quasars reported by Lusso & Risaliti (2016). We apply a Gaussian prior centred at the estimated value for L_{2500} , with a standard deviation of $\sigma = 0.3$ dex. In the case of X-ray-undetected sources, we do not include the AD emission in our SED modelling. As demonstrated below in Section 4.2.1, X-ray-undetected AGN are typically obscured by $N_{\rm H} > 10^{22}$ cm⁻². Hence, most or all of the

optical emission coming from the AD will be extinguished, and it is a reasonable assumption to associate all of the observed optical emission with the stellar population. We note that ≤ 15 per cent of the AGN with $N_{\rm H} > 10^{21.5}$ cm⁻² are expected to show signs of broad lines or a reddened AD in their optical spectra (e.g. Merloni et al. 2014); hence, not including the AD component in the SED modelling could cause a small fraction of the X-ray-undetected sources to have overestimated stellar masses. However, we do not consider this a major issue as the main objective of the SED modelling is to identify the IR AGNs. We summarize the different components and parameters of our SED modelling in Table 2. For more details on each model and SED-fitting implementation, we refer the interested reader to section 3 of Andonie et al. (2022). Fig. B1 in Appendix B shows examples of the best-fitting SED models for two 4MOST IR AGN sources in the XMM-LSS field.

We classify a source as an AGN only if the 8 – 1000 μ m AGN luminosity ($L_{AGN,IR}$) posterior distribution is well constrained, following the approach of Andonie et al. (2022, tabsee section 3.3 of that paper). We consider $L_{AGN,IR}$ to be well constrained (i.e. the source hosts an AGN) when the first percentile of log $L_{AGN,IR}/(\text{erg s}^{-1})$ is larger than 38.5 (i.e. > 0.5 dex from the lower limit of the prior; see Table 2). Similarly, we consider $L_{SF,IR}$ to be well constrained when the 1st percentile of log $L_{AGN,IR}/(\text{erg s}^{-1}) = 38.5$ and log $L_{SF,IR,1st percentile}/(\text{erg s}^{-1}) = 42.5$ do not have a redshift dependence and they are only an indication on whether the AGN or SF luminosity posterior distributions have a non-zero probability of taking the lowest possible value (i.e. having a value < 0.5 dex from the lower limit of the prior).

Throughout the text, to compute the median values and standard deviation of $L_{AGN,IR}$, $L_{SF,IR}$, and M_{\star} of different samples, we use POS-TERIOR STACKER.⁸ This code takes as input samples of posterior distributions and fits them using a Gaussian model (Baronchelli, Nandra & Buchner 2020), taking into account all of the information contained in the posterior distribution of each parameter to analyse the samples.

2.4 X-ray data, spectral constraints, and X-ray stacking

In this work, we use different X-ray data sets (*Chandra*; XMM– *Newton*) to perform a variety of analyses; see Table 1. Our primary

⁸See https://github.com/JohannesBuchner/PosteriorStacker/tree/main

X-ray data set is the *Chandra* data in the COSMOS field, which we use to demonstrate our obscured AGN selection approach. The XMM-LSS, Boötes, and ELAIS-S1 X-ray data sets are used solely to make predictions of the basic obscuration properties of the 4MOST IR AGN survey. Finally, the X-ray data from the *eROSITA* Final Equatorial-Depth Survey (eFEDS) are utilized as an equivalent of the 4MOST X-ray AGN survey to compare with the properties of the 4MOST IR AGN survey.

2.4.1 COSMOS survey

The deepest and most complete X-ray observations in COSMOS were carried out using *Chandra*, as part of the *Chandra* COSMOS-Legacy survey (Civano et al. 2016). This is a 4.6 Ms *Chandra* program that has observed the 2.2 deg² area of the COSMOS field, with a homogeneous exposure time of ~ 160 ks, yielding a 2–10 keV flux limit of $F_{2-10 \text{ keV}} \sim 1.9 \times 10^{-15} \text{ erg cm}^{-2} \text{ s}^{-1}$.

The properties of the X-ray sources in the COSMOS field have been extensively studied in previous works. In this work, we adopt the X-ray spectral fitting measurements from Laloux et al. (2023), who used a Bayesian approach and adopted a physically motivated torus model (UXCLUMPY, Buchner et al. 2019) to robustly constrain the physical parameters of all the X-ray-detected sources in COSMOS. They provide the 2 – 10 keV X-ray luminosity ($L_{2-10 \text{ keV}}$), the line of sight column density ($N_{\rm H}$), and the obscured fraction of each source ($f_{\rm obs}$; e.g. the probability of each source of being obscured by $N_{\rm H} > 10^{22} \text{ cm}^{-2}$).

2.4.2 Boötes, XMM-LSS, and ELAIS-S1 surveys

The Boötes, XMM-LSS, and ELAIS-S1 fields do not have X-ray observations as deep as COSMOS or the same level of published X-ray spectral constraints. However, they have available X-ray point source catalogues which are suitable for the analyses we want to perform; i.e. to provide an estimate of the X-ray luminosity and make predictions of the X-ray properties of the 4MOST IR AGN sample such as the fraction of X-ray-detected sources and the fraction of obscured sources with $N_{\rm H} > 10^{22}$ cm⁻². The X-ray luminosities are only used in our SED modelling to place a loose prior on the AD luminosity (see Section 2.3.2). Hence, the variety in the X-ray quality and information that we have for the various fields is not a significant limitation.

The Boötes field has been extensively observed by *Chandra* over more than 15 yr. Here, we use the catalogue of the Chandra Deep Wide-Field Survey from Masini et al. (2020), which combines all available *Chandra* observations, reaching a 2 – 7 keV limiting depth of 9×10^{-16} erg cm⁻² s⁻¹. Their catalogue provides estimates for the intrinsic X-ray luminosity and $N_{\rm H}$ measurements derived from hardness ratio (HR) analyses. The HR is defined as HR = (H – S)/(H + S), where H and S are the counts in the hard (2 – 7 keV) and soft (0.5 – 2 keV) bands, respectively. This method can provide an estimation of the $N_{\rm H}$, as obscuration mostly affects the soft X-ray band, while hard X-ray photons are able to penetrate larger column densities (for more details, see Section 6.2 of Masini et al. 2020).

The XMM-LSS field has been observed by XMM–Newton $(2 - 10 \text{ keV} \text{ limiting depth of } 3 \times 10^{-15} \text{ erg cm}^{-2} \text{ s}^{-1})$ and has a public X-ray source catalogue (Chiappetti et al. 2018), which contains information on the observed flux and count rates of each source at different bands. We estimate the observed X-ray luminosity of the sources using the provided observed flux in the 2 - 10 keV band. For this, we follow Alexander et al. (2003) and calculate

 $L_{2-10 \text{ keV}} = 4\pi d_{\text{L}}^2 f_{2-10 \text{ keV}} (1+z)^{\Gamma-2}$, where d_{L}^2 is the luminosity distance, $f_{2-10 \text{ keV}}$ is the observed flux in the 2 – 10 keV band, and Γ is the photon index which we set to $\Gamma = 1.8$. We note that the observed 2 – 10 keV X-ray luminosity provides a good estimate of the intrinsic X-ray luminosity as for the majority of the sources the absorption correction is small in the observed 2 – 10 keV band.⁹

The ELAIS-S1 field has also been observed by *XMM–Newton* as part of the XMM-SERVS Survey (Ni et al. 2021, 0.5 – 10 keV limiting depth of 1.3×10^{-14} erg cm⁻² s⁻¹), for which the X-ray point source catalogue contains the information on the observed flux of each source at different bands. Yan et al. (2023) studied the most obscured AGN in ELAIS-S1 and published the X-ray spectral constraints only for the heavily obscured AGN with $N_{\rm H} > 5 \times 10^{23}$ cm⁻². The $N_{\rm H}$ values for the rest of the sources were acquired by private communication. To estimate the X-ray luminosity, we follow the same approach as for the XMM-LSS field.

2.4.3 eROSITA Final Equatorial-Depth Survey

The eFEDS (Brunner et al. 2022) is a 140 deg² area observed by *eROSITA* as part of the performance verification phase ahead of the planned four years of *eROSITA* all-sky scanning operations. eFEDS contains 27 910 X-ray sources (out of which ~ 22 000 are AGN) detected down to 6.5×10^{-15} erg cm⁻² s⁻¹ in the 0.5 - 2 keV band. The objective of eFEDs was to observe a small region to the full depth of the all-sky *eROSITA* only performed about half of the planned all-sky observations (only 4–5 scans were completed, instead of the planned eight scans); hence, the eFEDS observations are about two times deeper than the final observations across the majority of the sky, and ~ 4 times deeper than the publicly available first-pass all-sky survey (Merloni et al. 2024).

In this work, we use the multiwavelength counterparts catalogue from Salvato et al. (2022) to obtain the optical positions and the photometric information of the eFEDS sources, which were originally taken from the Legacy Surveys DR 8 (LS DR8).¹⁰ In addition, we also use the AGN catalogue from Liu et al. (2022), which published the X-ray spectral fitting results of all the sources, such as the intrinsic $L_{2-10 \text{ keV}}$ and N_{H} .

2.4.4 X-ray stacking

To analyse the obscuration properties of the obscured IR AGN candidates that are undetected in the X-ray band, we performed X-ray stacking using the publicly available PYTHON code STACKFAST¹¹ (see also Ananna et al. 2024). In brief, STACKFAST is an X-ray stacking code that finds all of the *Chandra* observations of a given sample of objects and extracts the photons within the 90 per cent energy encircling radius ($r_{ee,90}$) around the position of each object.

⁹For example, assuming an absorbed power-law model, for an AGN with $N_{\rm H} = 10^{23}$ cm⁻² at $z \sim 1$, the observed X-ray luminosity would correspond to 90 per cent of the absorption-corrected X-ray luminosity.

¹⁰We note that in Appendix A, we mention that the LS handles optically extended sources in a way that is not useful for identifying IR AGN. However, all the eFEDS sources are X-ray AGN by selection, and most of them have an optically point-like morphology. Hence, using *WISE* photometry from the LS DR 8 is not an issue for the analysis performed in the eFEDS (see Section 4.3). ¹¹The STACKFAST code is available in the following websites: https://www.tonima-ananna.com/research and https://gitfront.io/r/user-5016120/9154498a888cfd94976fc57b51b447460c69b268/StackFast/.

To calculate the net photons of each object, STACKFAST adopts a background with annular shape of radii (1.3–6) $r_{ee,90}$. It then calculates fluxes for each individual object and the mean stacked flux for the entire sample in the soft (0.5–2 keV) and hard (2–7 keV) X-ray band.

3 4MOST IR AGN SURVEY DESIGN

This section describes the steps performed to define the 4MOST IR AGN survey (Section 3.1) and demonstrates two key factors: (1) the effectiveness of the optical-to-MIR colour cut to distinguish between obscured and unobscured AGN (Section 3.2), and (2) the reliability of our AGN identification approach (Section 3.3).

3.1 The 4MOST IR AGN survey selection function

The 4MOST IR AGN survey selection function has several steps, which are illustrated in Fig. 1. Every decision made during the selection has been previously justified in Section 2.2, or will be well justified and validated later in this section using detailed UV-to-FIR SED template fitting and X-ray spectral constraints. We first find all the matches between the LS DR10 and the IR AGN catalogue, following Section 2.2 using the unWISE catalogue (see Section 3.3). Given the aims of our survey and the properties of the 4MOST instrument, we choose a magnitude limit of $r_{AB} < 22.8$ in order to achieve at least our minimum goals (i.e. a measured spectroscopic redshift; see Section 4.1), for the majority of sources. Then, in order to select the most obscured AGN, we apply the optical to MIR colour cut of $r - W^2 > 5.9 \text{ mag}$ (Hickox et al. 2007), following Section 3.2. This optical-to-MIR colour cut also delivers an obscured sample that complements the properties of the 4MOST X-ray AGN survey (see Section 4.3 and Fig. 12). At this point, we have $\sim 470\,000$ IR AGN candidates. As we need the IR AGN survey to spatially overlap with the 4MOST X-ray survey, we limit our catalogue to the 4MOST X-ray survey footprint, reaching $\sim 212\,000$ sources, applying a $r_{AB} < 22.1$ cut for those that lie in the 'IR AGN wide' footprint (see below).

The 4MOST IR AGN survey has two components; see the spatial footprint of the survey in Fig. 2. A deep component, which will target $\approx 30\,000$ sources down to $r_{AB} < 22.8$ over $\sim 850 \text{ deg}^2$ with more extensive deeper multiwavelength photometry, and a wide component, which will target $\approx 182\,000$ sources down to $r_{AB} < 22.1$ over $\sim 9000 \text{ deg}^2$. The deep component follows the footprints of two extragalactic surveys: the Wide Area Vista Extragalactic Survey (WAVES, Driver et al. 2019), which is confined to the footprint of the Kilo-Degree Survey (de Jong et al. 2013) and the VISTA Kilo-Degree Infrared Galaxy Survey (Edge et al. 2013); and the XMM-LSS field as defined by the LSST Deep Drilling Fields (DDF, Brandt et al. 2018). This ensures that our deep survey will have complementary NIR photometry over the entire footprint, and additional MIR and FIR photometry in the XMM-LSS area. Fig. 2 also shows the overlap of the 4MOST IR AGN survey with other key public spectroscopic and photometric surveys. Notably, we highlight the overlap of $\sim 2500 \text{ deg}^2$ between the 4MOST IR AGN survey and DESI, including the entire equatorial IR AGN deep field, which will result in exceptionally deep and extensive optical spectroscopic data across this region.

We note that full UV–FIR SED modelling would be the most effective way to select obscured IR AGN; however, we cannot use this approach to reliably construct the 4MOST IR AGN survey because the required deep multiwavelength observations out to FIR wavelengths only exist over a small fraction of the sky (see Fig. 2



IR AGN candidates in unWISE

Figure 1. 4MOST IR AGN survey selection function. We first select all the IR AGN that meet the Assef et al. (2018) criteria using the unWISE catalogue with -80 < Dec < 6, and find all the possible matches within 2 arcsec of targets in the Legacy Survey DR10 (LS10). When one unWISE source has multiple optical counterparts in the LS10, we pick the source with the largest *WISE W2* flux in the LS10 catalogue. To select the obscured IR AGN, we apply the colour cut $r - W2_{\text{unWISE}} \ge 5.9$ mag, and then restrict our sources to $r_{\text{AB}} < 22.8$, which corresponds to the optical limit of the 4MOST IR AGN survey. Given that our survey is designed to complement the 4MOST X-ray survey, we limit the catalogue to the 4MOST *eROSITA* footprint. Finally, we define the two survey components within this footprint: IR AGN deep, which will target sources down to $r_{\text{AB}} = 22.8$ in the WAVES and XMM-LSS (DDF area) regions (~ 850 deg²), and IR AGN wide, which will target sources for $r_{\text{AB}} = 22.1$ across the rest of the survey footprint (~ 9000 deg²; see Fig. 2).

for the *Herschel* coverage). Instead we adopt *WISE*, an all-sky MIR survey, which offers us the widest possible areal coverage and a reliable obscured AGN selection strategy.

3.2 Quantifying the r - W2 colour as a proxy for AGN obscuration

To select obscured AGN, the key objective of the 4MOST IR AGN survey, we require a diagnostic sensitive to obscuration. Hickox et al. (2007) presented a simple optical-to-MIR colour cut to select obscured AGN. Unlike the MIR emission, the optical emission is very sensitive to obscuration and gets extinguished at relatively low



Figure 2. Aitoff projection of the 4MOST IR AGN survey sources in equatorial coordinates (J2000). The light grey highlighted region indicates the IR AGN wide component ($r_{AB} < 22.1$), and the dark grey highlighted region indicates the IR AGN deep component ($r_{AB} < 22.8$). The full 4MOST IR AGN survey footprint has been observed by *eROSITA* and follows that of the 4MOST X-ray AGN survey. We also show some multiwavelength surveys overlapping with the 4MOST footprint. Red regions show fields with *Herschel* coverage, asterisks show fields with *Spitzer* MIPS 24μ m coverage, crosses show fields with deeper X-ray coverage from *Chandra* or *XMM–Newton*, and the purple open region indicates the coverage map of the DESI survey which partially overlaps (~ 2500 deg²) with our 4MOST IR AGN survey, including all of the equatorial IR AGN deep field.

obscuration levels. Hence, the optical-to-MIR colour is expected to vary with different amounts of obscuration. Hickox et al. (2007) proposed to use the colour r - [4.5] as a proxy for AGN obscuration, where [4.5] correspond to the *Spitzer* IRAC 2 channel at 4.5 μ m. Based on the observed bimodal distribution of r - [4.5] colour as well as the results of X-ray stacking analyses, they demonstrated that the boundary $r - [4.5] \gtrsim 6.1$ mag selects obscured AGN with an average $N_{\rm H} \approx 3 \times 10^{22}$ cm⁻². A decade later, Hickox et al. (2017) extended these analyses by studying the SED and spectral properties of SDSS quasars as a function of the optical to MIR colour, finding that obscured and unobscured quasars have remarkably similar UV–MIR SEDs with the major difference being clear UV– optical extinction in the obscured quasars. However, they did not explore the dependency of r - [4.5] with the absorbing column density.

In this paper, we extend the Hickox et al. (2007, 2017) analyses to explore the relationship between r - [4.5] and AGN obscuration using the X-ray spectral constraints in the COSMOS field, by analysing the line-of-sight column density $(N_{\rm H})$, the obscured fraction (f_{obs} , see Section 2.4 for details), and the optical extinction towards the AD. For this analysis, we performed UV-to-FIR SED fitting of all the X-ray AGN in COSMOS to constrain the AD reddening $E(B - V)_{AD}$ and the IR AGN luminosity, following the procedure described in Section 2.3.2. Since we use WISE W2 4.6 μ m to construct the 4MOST IR AGN survey instead of Spitzer/IRAC 2 (since there is Spitzer/IRAC coverage over only a small fraction of the sky), we also extend these analyses using r - W2 (e.g. DiPompeo et al. 2015; Petter et al. 2023). However, Spitzer/IRAC is much deeper than WISE in the COSMOS field, providing better source statistics. We note that in our analyses here, we adopt r - [4.5] > 5.9 magas the (predominantly) obscured AGN threshold to align with the

definition of our 4MOST IR AGN survey. This r - [4.5] cut is slightly less conservative than the $r - [4.5] \ge 6.1$ mag threshold adopted in Hickox et al. (2007). However, we note that this change does not quantitatively change the results presented in this paper and selects more sources in the interesting obscured–unobscured threshold where we may identify rare systems such as red and extremely red quasars (Ross et al. 2015; Hamann et al. 2017; Fawcett et al. 2023).

Fig. 3 shows the optical-to-MIR colour against redshift for the X-ray AGN in the COSMOS field. The top panel plots r - [4.5] for the sources detected by Spitzer/IRAC 2 and the bottom panel plots r - W2 for the sources with W2 detections in the unWISE catalogue. We find consistent results when using Spitzer/IRAC 2 and W2.¹² In both cases, ~ 65 per cent of the Xray AGN with $r - [4.5]/W2 \ge 5.9$ mag colours are obscured by $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$, while only $\sim 30 - 35$ per cent of the X-ray AGN with $r - [4.5]/W^2 < 5.9$ mag colours have $N_{\rm H} > 10^{22}$ cm⁻². This is better illustrated in Fig. 4, which shows the distribution of $\log N_{\rm H}$ for sources with $r - [4.5]/W^2 < 5.9 \text{ mag}, 5.9 \text{ mag} \le r - [4.5]/W^2 < 5.9 \text{ mag}$ 7.5 mag, and $r - [4.5]/W2 \ge 7.5$ mag, and also plots vertical lines at the median $\log N_{\rm H}$ of each colour distribution. A progression can clearly be seen where AGNs with redder r - [4.5]/W2colours have larger $N_{\rm H}$. We find that the $r - [4.5]/W^2 < 5.9$ mag colour cut is more efficient in identifying X-ray unobscured AGN with $N_{\rm H} < 10^{22} \,{\rm cm}^{-2}$ than the 5.9 mag $< r - [4.5]/W^2 < 7.5$ mag colour cut. Consequently, the 5.9 mag $\leq r - [4.5]/W^2 < 7.5$ mag cut is much more effective at identifying X-ray obscured AGN

¹²From now on, we use the expression r - [4.5]/W2 to refer to r - [4.5] and r - W2 at the same time.



Figure 3. Optical-to-MIR colour against redshift, colour-coded by the lineof-sight logarithmic column density (log $N_{\rm H}$) for all the X-ray AGN with X-ray spectral constraints in the Chandra COSMOS field. The top panel plots the r - [4.5] colour for all the sources detected by Spitzer/IRAC 2 at 4.5 μ m, and the bottom panel plots the r – W2 colour for all the sources with a W2 detection in the unWISE catalogue. The green line shows r – [4.5]/W2 = 5.9 mag, the colour cut adopted in this work to separate obscured from unobscured AGN, based on Hickox et al. (2007). The majority of the AGN with $N_{\rm H} \ge 10^{22} \,{\rm cm}^{-2}$ have $r - [4.5]/W^2 \ge 5.9$ colour.

with $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$. Notably, the majority (~ 75 per cent) of the AGNs with $r - [4.5]/W2 \ge 7.5$ mag are obscured by $N_{\rm H} >$ 10^{22} cm⁻². We find that the median obscuration of sources with $r - 10^{22}$ cm⁻². [4.5]/W2 < 5.9 mag is $\log N_{\text{H,r}-4.5/W2<5.9}/(\text{cm}^{-2}) \approx 21.5$, while the median obscuration of sources with $r - [4.5]/W^2 \ge 5.9 \text{ mag}$ is log $N_{\rm H,r-4.5/W2 \ge 5.9}/(\rm cm^{-2}) \approx 22.5$, consistent with Hickox et al. (2007). Similar median values are found when we only consider the X-ray AGNs that are also identified as IR AGNs from our broad-band SED fitting (log $N_{\rm H,r-4.5/W2<5.9}/(\rm cm^{-2}) \approx 21.3$ and $\log N_{\rm H,r-4.5/W2 \ge 5.9}/(\rm cm^{-2}) \approx 22.6$).

Fig. 5 depicts the variation of the X-ray column density $N_{\rm H}$ (panel b), the obscured fraction f_{obs} (panel c), and the AD reddening $E(B - V)_{AB}$ (panel d) as a function of $r - [4.5]/W^2$ colour. We perform this analysis for all X-ray AGNs, and separately examine the subset that also has an IR AGN component identified by our SED fitting. We can see a clear positive trend between the two Xray obscuration features and $E(B - V)_{AB}$ with the optical-to-MIR colour. Panels (b) and (c) indicate that the wide majority of the objects with $r - [4.5]/W2 \gtrsim 6.5$ mag are obscured by $N_{\rm H} > 10^{22}$ cm⁻². For $r = [4.5]/W^2 \gtrsim 5 \text{ mag}, N_{\rm H}$ and $f_{\rm obs}$ systematically increase with the optical-to-MIR colour reaching a constant value at $r - [4.5]/W2 \approx$ 8 mag. These trends suggest that when $N_{\rm H}$ increases, the amount of dust, and hence, dust reddening in the system also increases, progressively extinguishing the optical emission, which is also



Figure 4. Distributions of the logarithmic column density, $\log N_{\rm H}$ (cm⁻²), for the X-ray AGN in COSMOS with $r - [4.5]/W^2 < 5.9$ mag (dashed blue), $5.9 \text{ mag} \le r - [4.5]/W^2 < 7.5 \text{ mag}$ (solid green), and $r - [4.5]/W^2 \ge 1000$ 7.5 mag (dotted red). The vertical lines show the median $\log N_{\rm H} \, ({\rm cm}^{-2})$ of each colour distribution, following the same colour code and line style as the histograms. The top panel plots the sources detected by Spitzer/IRAC, and the bottom panel plots the sources detected in the unWISE catalogue. The $r - [4.5]/W2 \ge 5.9$ mag colour cut predominantly selects obscured

observed in panel (d). At $r - [4.5]/W2 \gtrsim 7.5$ mag, the $N_{\rm H}$, $f_{\rm obs}$, and $E(B - V)_{AB}$ reach a constant value probably due to the AGN emission in the optical waveband being completely extinguished at those high obscuration levels. Indeed, the optical wavelength is expected to be fully extinguished at $A_V \gtrsim 5$ (e.g. Schnorr-Müller et al. 2016; Burtscher et al. 2016), which is equivalent to $N_{\rm H} \gtrsim 10^{22} \, {\rm cm}^{-2}$ (assuming the gas-to-dust ratio observed in our galaxy Predehl & Schmitt 1995).¹³ Hence, at $r - [4.5]/W2 \gtrsim 7.5$ mag, where the obscured fraction is > 80 - 90 per cent ($N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$),¹⁴ we expect the optical emission to be dominated by host galaxy light, and hence, the $r - [4.5]/W^2$ colour will saturate. In addition, at those very high levels of obscuration, the MIR emission at $\sim 4.5 \mu m$ can also be absorbed, reaching a regime where the value of the $r - [4.5]/W^2$ colour will only indicate whether the source is obscured or not, but will not give an estimate of the exact level of obscuration. Notably, we find no strong dependence between redshift

AGN.

¹³We note that we would need high S/N optical spectroscopy, as the one that will be provided by 4MOST, to constrain the AD reddening beyond $E(B-V)_{\rm AD} \gtrsim 0.7$).

¹⁴We note that, at $r - W^2 > 7.5$ mag, the obscured fraction calculated using the f_{obs} parameter is larger than the value obtained when just calculating the fraction of sources with $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$ (~ 75 per cent), which is not surprising as these quantities were calculated using different approaches.



Figure 5. Different obscuration parameters and redshift as a function of the r - [4.5] (and $r - W^2$) colour. The black circles represent all the X-ray AGN in the COSMOS field, and the red stars are the X-ray AGN that are also identified as IR AGN by the UV-to-FIR SED fitting. The error bars in the x-axis represent the size of the bin in all the panels. The y-axis values are the median value of the given obscuration parameter in each r - [4.5] colour bin, and the error bars represent the 95 per cent confidence interval of the median of each parameter, calculated using bootstrap resampling within each colour bin. The grey shaded area in each panel illustrates the relationship between each obscuration parameter as a function of r - W2 instead (top y-axis). Panel (a) shows the number of sources per r - [4.5] (black bar) and r - W2(grey bar) colour bin; panel (b) shows the line-of-sight X-ray column density $(N_{\rm H})$ in logarithmic scale (the median was also calculated in logarithmic space); panel (c) shows the obscured AGN fraction f_{obs} calculated from the $N_{\rm H}$ posterior distribution (see Section 2.4 for details); panel (d) shows the AD reddening $E(B - V)_{AD}$ calculated from the UV-to-FIR SED fitting; and panel (e) shows the median and standard deviation of the redshift distribution of each r - [4.5] colour bin.

and r - [4.5]/W2 colour (see panel e of Fig. 5); hence, the positive trend between obscuration and r[4.5]/W2 is not driven by redshift effects.

This subsection demonstrated that the r - [4.5] and r - W2 colours are good proxies of AGN obscuration. Particularly, the correspondence between the X-ray obscuration features and r - [4.5]/W2 is notably impressive, given that they are fully independent quantities. In addition, as will be demonstrated in the following section, the r - W2 > 5.9 cut also provides a good balance between obscuration properties and sample size. We note that the results of this section are also consistent with the X-ray properties of extremely red objects, which are typically selected using optical to NIR colours (e.g. r - K), and have also been shown to be X-ray obscured (e.g. Alexander et al. 2002; Brusa et al. 2002, 2005, 2010; Mainieri et al. 2005).

3.3 AGN reliability of our obscured IR AGN selection

This subsection demonstrates the effectiveness of the A18 R90 colour selection (see equation 1 and Fig. A1 of Appendix A) to reliably identify AGN. The A18 R90 AGN selection is based on W1 - W2 colours (see Fig. A1) with 3σ and 5σ detections in W1 and W2, respectively. As demonstrated later in this work (see Section 4.2 and Appendix A) and also in previous studies (e.g. Barrows et al. 2021), the A18 R90 selection provides a robust AGN sample, spanning across a wide AGN luminosity and redshift range.

Here, we use detailed UV-to-FIR SED fitting (see Section 2.3.2) to investigate the performance of the A18 R90 selection for the unWISE catalogue. For this, we identify A18 R90 IR AGN candidates in four deep, well-observed fields in the sky with UV-to-FIR photometric coverage described in Section 2.3.1: the COSMOS, Boötes, ELAIS-S1, and XMM-LSS fields. We then calculate the AGN reliability of the A18 R90 selection by assuming that the reliable AGN are those identified by our SED template fitting approach described in Section 2.3.2.

Fig. 6 shows the AGN reliability of the A18 R90 selection as a function of r - W2 for the unWISE catalogue,¹⁵ for $r_{AB} < 22.1$ and $r_{AB} < 22.8$, which are the depths of our wide and deep IR AGN survey components, respectively. The AGN reliability as a function of the r - W2 colour is important as we apply the colour cut $r - W^2 > 5.9$ mag to identify the obscured IR AGN (see Section 3.2). We find that at r - W2 < 4 mag colours, the sample is highly contaminated by quiescent, massive galaxies (see Fig. A4 in Appendix A2) with an AGN reliability of only ≈ 20 per cent. However, at $r - W2 \ge 4$ mag, the AGN reliability drastically improves to > 80 per cent. At $r - W2 \ge 5.9$ mag, which is our obscured AGN selection, the A18 R90 selection provides a clean AGN sample with an AGN reliability of 87 per cent for $r_{AB} < 22.1$ and 80 per cent for $r_{AB} < 22.8$. Table 3 provides a summary of the AGN reliability and number density of the A18 R90 selection applied to unWISE photometry.

We conclude that our obscured IR AGN selection (i.e. A18 R90 selection plus $r - W2 \ge 5.9$ mag) applied to unWISE photometry provides a robust and clean AGN sample to define the 4MOST IR

¹⁵We note that A18 used UV-to-MIR SED fitting to arrive at their 90 per cent AGN reliability threshold using AllWISE photometry (see Assef et al. 2013, for details on their SED-fitting approach). In our work, we also independently assess the reliability of the approach using other deeper *WISE* photometry, such as unWISE and the LSDR10, and we use a more sophisticated SED modelling which also includes the FIR band.



Figure 6. AGN reliability as a function of the r - W2 colour for the AGN candidates selected by the Assef et al. (2018) R90 AGN colour selection and using unWISE photometry down to $r_{AB} < 22.1$ mag (black circles) and $r_{AB} < 22.8$ mag (red stars). The AGN reliability corresponds to the fraction of colour-selected AGN that are also identified as AGN by our SED fitting. The error bar sizes in the *x*-axis represent the half width of the r - W2 bins, while the *y*-axis error bars correspond to the 1σ binomial error. The widths of the r - W2 bins are 0.5 mag for all bins except the lowest (r - W2 = 2 - 4 mag) and highest (r - W2 = 7.5 - 9 mag) values, which have bin widths of 2 and 1.6 mag, respectively, since there are not enough sources with such extremes r - W2 colours. The green dotted vertical line marks r - W2 = 5.9 mag, which is our threshold to separate obscured from unobscured IR AGN candidates. The upper panel shows the number of sources per r - W2 bin down to $r_{AB} < 22.1$ mag (red bar) and $r_{AB} < 22.8$ mag (black bar).

Table 3. AGN reliability (*R*) and AGN number density per deg² (*N*) of the AGN candidates identified by the A18 R90 selection using unWISE photometry. The AGN reliability is calculated based on the UV-to-FIR SED-fitting results. We also report the values for the obscured Assef et al. (2018) AGN with $r - W2 \ge 5.9$ mag colours (bottom rows).

	R (per cent)	$N (\text{deg}^{-2})$
$r_{\rm AB} < 22.8$	68	140
$r_{\rm AB} < 22.1$	69	125
$r_{\rm AB} < 22.8$	80	32
$r_{\rm AB} < 22.1$	87	18
	$r_{AB} < 22.8$ $r_{AB} < 22.1$ $r_{AB} < 22.8$ $r_{AB} < 22.1$	R (per cent) $r_{AB} < 22.8$ 68 $r_{AB} < 22.1$ 69 $r_{AB} < 22.8$ 80 $r_{AB} < 22.1$ 87

AGN survey. For the interested reader, Appendix A presents a more exhaustive analysis of the efficiency of the A18 R90 selection applied to different *WISE* catalogues, its source of contamination, and also demonstrates that the unWISE photometry provides the best option to construct the 4MOST IR AGN sample.

4 4MOST IR AGN SURVEY: PREDICTIONS AND OBJECTIVES

This section presents our predictions for the properties of the targets in the 4MOST IR AGN survey and our future research plans. Our final survey consists of $\approx 212\,000$ obscured MIR-selected AGN candidates and has two components: (1) the IR AGN wide component, which will observe $\approx 182\,000$ mostly obscured IR AGN over an area of $\sim 9000 \text{ deg}^2$ down to $r_{AB} = 22.1$, and (2) the IR AGN deep component, which will observe $\approx 30\,000$ obscured IR AGN over a smaller but extensively observed area of $\sim 850 \text{ deg}^2$ down to $r_{AB} = 22.8$. Section 4.1 presents predictions for the optical spectral properties of the 4MOST IR AGN survey; Section 4.2 presents predictions for the AGN and host galaxy properties of 4MOST IR AGN sample calculated in deep extensively observed regions described in Section 2.3; Section 4.3 demonstrates that the 4MOST IR AGN survey complements the 4MOST X-ray AGN survey by targeting AGN with different properties; and in Section 4.4, we present the scientific goals and future research plans we expect to execute with the 4MOST IR AGN survey.

4.1 Spectroscopic predictions for the 4MOST IR AGN survey

To inform our target selection (particularly our optical magnitude limits) and to test that the 4MOST spectra will lead to scientifically useful spectra, we simulated targets using the 4MOST PAQS spectral simulator.¹⁶ This tool simulates the optical spectra of individual targets and calculates the expected exposure times using the 4MOST exposure time calculator. We take the following steps to produce simulated spectra with appropriate properties:

(i) Randomly select targets for simulation from our target catalogue.¹⁷ These objects have *r*-band magnitude measurements, and we choose targets so that the magnitude distributions match those found in detailed studies of the deep fields (see Sections 2.3.1 and 4.2). We also assign redshifts to these targets so that the distribution across the redshift–magnitude plane of simulated targets will be similar to the likely observed distributions.

(ii) Based on the distribution of the obscured fraction as a function of r-band magnitude calculated using the predictions in the deep fields (see Section 4.2), we randomly assign a Type 1 or Type 2 quasar template to each target. The Type 1 template was produced from SDSS sources observed as part of the SPectroscopic IDentifcation of ERosita Sources (SPIDERS) survey (Comparat et al. 2020). The Type 2 template is a composite of narrow-line AGNs meeting our target-selection approach found in the DESI early data release (DESI Collaboration 2024).

(iii) Simulate an observed spectrum with the PAQS spectral simulator for each target based on its redshift, magnitude, and spectral template (see Fig. 7). This tool also calculates an associated exposure time which is adjusted automatically to produce a signal-to-noise (S/N) according to rules defined specifically for the IR AGN subsurveys.

(iv) Apply the redshift and classification algorithm of 4MOST, which uses principal component analysis (PCA) template fitting (xPCA; Krogager in preparation), to determine a best-fitting redshift and object type to each simulated spectrum.

Based on these simulated optical spectra and template fitting results, we can assess the full results. For the wide survey simulations

¹⁶https://github.com/jkrogager/simpaqs

¹⁷10 000 for each of the two subsurveys.



Figure 7. Randomly selected simulated optical spectra with smoothing applied at *r*-band magnitudes of 21.5 (top panel), 22.1 (middle panel), and 22.5 (bottom panel). Major emission lines are indicated by vertical dashed grey lines. Spectra at a range of redshifts and with both template types are shown.



Figure 8. Redshifts recovered from simulated spectra as a function of r-band magnitude and redshift. The central panel shows the fraction of targets with a fitted redshift that matches the simulation input redshift compared to total targets within each bin. Darker colours represent a higher fraction of successful results and empty bins contain no simulated targets. Histograms show these results as a function of each parameter separately: total distribution of targets (filled grey bins) and distribution of targets with redshifts recovered (solid black lines). This figure shows results for the deep survey – results for the wide survey are very similar below the r-band magnitude cut-off.

(mean optical magnitude of $r_{AB} \approx 21.4$), the mean and maximum exposure times are 14.5 and 37.2 min, respectively, yielding a total exposure time for all targets of 44 000 h. For the deep survey (mean optical magnitude of $r_{AB} \approx 21.8$), the mean and maximum exposure times of the simulated targets are 29.4 and 96.4 min, respectively, which results in a total exposure time of 15 000 hr.

The minimum science goal of the IR AGN survey is to measure optical spectroscopic redshifts (see Section 4.4). We define a redshift as successfully recovered if the redshift returned by xPCA is close to the redshift input to the simulation, i.e. $\frac{|z_{sim} - z_{fir}|}{1 + z_{sim}} < 0.05$.¹⁸ Fig. 8 shows that the redshift recovery rate varies with redshift but not significantly with magnitude, as also found for the 4MOST-ORCHIDSS survey (Duncan et al. 2023). This can be due to the fact that the optical spectra of AGN typically have high equivalent width emission lines, which can be detected even in optically faint sources, as shown in Fig. 7. Additionally, in our survey, the exposure time varies with magnitude in order to obtain a consistent S/N [i.e. fainter sources are observed for longer, as described in Section 4.1, point (iii)]. With all redshifts included the total recovery rate is \approx 73 per cent (\approx 72 per cent) for the wide (deep) subsurvey. However, after removing the 'redshift desert' of 0.75 < z < 1.25, where the observed optical spectra often contain only a single emission line ([OII] λ 3727Å) that we can use to measure redshifts, the recovery rate increases to ≈ 80 per cent (≈ 82 per cent). We expect the total redshift recovery rate to improve with developments in the 4MOST pipelines (particularly the inclusion of photometric redshift priors, which have been shown to increase the performance significantly; Krogager et al., in preparation) and visual inspection (VI) analysis as the project moves forward. Quasars in particular can be difficult for general pipelines, and large improvements can be made using VI and combining the pipeline with additional 'afterburner' identification approaches (e.g. Alexander et al. 2023; Chaussidon et al. 2023). A simple check of simulated sources finds that in cases where the pipeline assigns the wrong redshift, the correct redshift can be easily identified with VI in many cases. For example, common misidentifications such as $[O II] \rightarrow Ly\alpha$, Mg II $\rightarrow Ly\alpha$, and $[O II] \rightarrow$ [O III] produce a significant fraction of wrongly assigned redshifts, depending on the strength of these emission lines.

Expanded analyses will involve detection and fitting of specific emission lines, with the aim of exploring the accretion properties of the targets. Fig. 7 shows simulated spectra, in which several important AGN emission lines are visible such as C IV, C III], Mg II, [O II] λ 3727Å, H β , [O III] λ 5008Å, and H α . Continuum emission is also clearly detectable. Full fitting of these simulated emission lines would not be informative because their specific properties depend on the input template and would therefore not provide an idea of expected results beyond the original assumptions adopted in the simulations.

Based on the results of this section, we are confident that, when strong emission lines are present (which is likely to be the case except for very extreme and unusual AGN), we will measure the redshifts and emission-line fluxes for the majority of 4MOST IR AGN survey targets. Furthermore, with sufficient S/N, we will also be able to perform more detailed optical spectral fitting and analyse further properties, such as the examples described in Section 4.4. With this unprecedentedly large number of optical spectra we will perform, for the first time, large statistical population analyses of obscured IR AGNs and quasars.

4.2 Photometric predictions for the 4MOST IR AGN target properties

This subsection explores the AGN and host galaxy properties of the obscured IR AGN candidates (i.e. the sources identified by our obscured IR AGN selection based on their *WISE* and r - W2 colours; see Section 3.3) in the deep fields. These properties will be used to predict the AGN and host galaxy properties of both components of the 4MOST IR AGN survey: IR AGN deep ($r_{AB} < 22.8$) and IR AGN wide ($r_{AB} < 22.1$). For this analysis, we apply our obscured IR AGN selection to the targets in the COSMOS, Boötes, XMM-LSS, and ELAIS-S1 fields, where we have the UV-to-FIR SED fitting (see Section 2.3.2) and X-ray observations from Chandra or XMM-Newton to investigate the X-ray obscuration properties. Our 4MOST IR AGN selection identifies 762 obscured IR AGN candidates with $r_{AB} < 22.8$ in those fields, from which 653 sources (85 per cent) have a high-quality spectroscopic or photometric redshift estimate and sufficient UV-to-FIR photometry to perform SED fitting; i.e. they are detected by at 24 μ m by Spitzer MIPS and have at least an FIR flux upper limit from Herschel, which is crucial to constrain the dust-obscured AGN and star formation components.

4.2.1 AGN properties: luminosity and obscuration

Fig. 9 shows the 8 – 1000 μ m AGN luminosity ($L_{AGN,IR}$) versus the redshift of the sample. As presented in Section 3.3, the SED fitting identifies an AGN in ~ 80 per cent and ~ 87 per cent of the obscured IR AGN candidates down to $r_{AB} < 22.8$ and < 22.1, respectively. For the sources which are not identified as AGN, the SED fitting could only provide an upper limit on their AGN luminosity. We note that the SED fitting can fail to identify an AGN in the case of low-luminosity AGN, in high SFR systems where the host-galaxy emission dilutes the emission from the AGN, or when there is poor photometric coverage.

¹⁸The specific choice in threshold makes little difference below ~ 0.1 .



Figure 9. The logarithmic 8 – 1000 μ m AGN luminosity ($L_{AGN,IR}$) against the redshift for sources that meet the 4MOST IR AGN selection for the wide survey (top panel) and deep survey (bottom panel) in the deep fields (black circles). The red stars are the IR AGN that are also X-ray AGN detected by *Chandra* or *XMM–Newton*. The $L_{AGN,IR}$ upper limits correspond to the 99th percentile of the $L_{AGN,IR}$ posterior distribution for the AGN candidates without a well-constrained AGN component in their SED. The histograms on the top and right sides of each figure show the redshift and $L_{AGN,IR}$ distributions of the IR AGNs (black solid) and the IR AGNs detected in the X-ray band (red dotted).

We find that the redshift of the obscured IR AGN candidates ranges $z \approx 0 - 3.5$, with median values of $z \approx 0.8$ and ≈ 1 for $r_{AB} < 22.1$ and < 22.8, respectively. In addition, we observe that both samples have very similar AGN luminosities: the IR AGN luminosity ranges $\log L_{AGN,IR}/(\exp s^{-1}) = 43.5 - 47$, with a median value of $\log L_{AGN,IR}/(\exp s^{-1}) = 45.1 \pm 0.8 (45 \pm 0.9)$ down to $r_{AB} < 22.1$ (22.8), and ≈ 55 per cent of the sources are IR quasars with an AGN luminosity $L_{AGN,IR} > 10^{45} \exp s^{-1}$. Based on these results, we predict that the 4MOST IR AGN deep and wide surveys will observe ~ 16500 and ~ 100000 IR quasars, respectively. To statistically test whether the IR AGN deep and wide samples have different redshift and AGN luminosity properties, we measure the *p*-value of the distributions using a log rank test, which also accounts for censored data (i.e. upper limits). For this, we use the PYTHON package LIFELINES.¹⁹ We find that the *p*-value of the $L_{AGN,IR}$ distributions is ≈ 0.6 , indicating that there are not significant differences between both samples. In the case of the redshift distributions, we find that the *p*-value between the wide and deep samples is ≈ 0.04 , suggesting stronger statistical differences, which is expected given that the sources in the IR AGN deep sample generally have larger redshifts.

Fig. 9 also shows the obscured IR AGN candidates that are detected in the X-ray band. We find that $\sim 45-50$ per cent of the sample are detected by Chandra or XMM-Newton down to $f_{0.5-10\text{keV}} \approx (0.09 - 1) \times 10^{-14} \text{ erg cm}^{-2} \text{ s}^{-1}$, while the other ~ 50-55 per cent are undetected. If we apply a simple X-ray flux cut using the eFEDS limiting depth, we find that only ~ 20 per cent of the sample would be detected by *eROSITA* down to $f_{0.5-2 \text{ keV}} = 1.5 \times$ 10^{-15} erg cm⁻² s⁻¹, consistent with the results found in Section 4.3. We find that the X-ray-detected IR AGN have consistent redshift and $L_{AGN IR}$ distributions with the obscured IR AGN candidates: the *p*-values of the redshift distributions and $L_{AGN,IR}$ distributions are always > 0.8. In addition, the COSMOS, Boötes, and ELAIS-S1 fields have measurements of the line-of-sight column density $N_{\rm H}$. We find that 83 out of 120 obscured IR AGN candidates that are detected in the X-ray band and also have N_H constraints are obscured by $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$, corresponding to ~ 70 per cent of the X-ray-detected sample. This value is in quantitative agreement with Section 3.2, where it was found that \sim 66 per cent of the X-ray AGN in the COSMOS field with $r - W2 \ge 5.9$ mag are obscured by $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$.

To assess the level of obscuration for the X-ray-undetected obscured IR AGN candidates, we follow the same approach as Andonie et al. (2022) and stack all the X-ray-undetected systems in the Boötes and COSMOS fields using STACKFAST (see Section 2.4.4 for details). We only perform this analysis in fields with *Chandra* observations since the background emission of the observations is fairly low, which makes it possible to obtain reliable background-subtracted count rates. We note that in the stacking, we also include the X-ray-undetected systems that do not have enough photometry to perform SED fitting.

Table 4 shows the stacking results in the soft and hard X-ray bands for the X-ray-undetected obscured IR AGN candidates in COSMOS, Boötes, and COMOS+ Boötes combined. In all three cases, we detect a strong signal in both X-ray bands. We also calculate the HR of the samples to estimate the level of obscuration for the Xray-undetected sources. Fig. 10 shows the HR against the median redshift of each sample, which is calculated only for the sources with either spectroscopic or photometric redshift estimates (most sources without photometry do not have redshift estimates). The plot also shows the expected change in the HR with redshift for a fixed column density and intrinsic X-ray spectral slope ($\Gamma = 1.8$) calculated using PIMMS v4.12d.²⁰ The HR suggests that the majority of the X-ray-undetected obscured IR AGN candidates are obscured by $N_{\rm H} > 10^{22} \, {\rm cm}^{-2}$, which is consistent with that expected given

¹⁹See https://pypi.org/project/lifelines/

²⁰See https://cxc.harvard.edu/toolkit/pimms.jsp. We choose to use the Chandra-Cycle 14 response files as that corresponds to the period when the majority of the *Chandra* observations for the *Chandra* COSMOS Legacy survey were taken. The majority of the *Chandra* observations in the Boötes field were taken in Chandra-Cycle 18, but we found that the *Chandra* response did not change significantly between Cycles 14 and 18.

Sample	Ν	$\frac{S}{(counts s^{-1})}$	H (counts s^{-1})	HR	Z
(1)	(2)	(3)	(4)	(5)	
COSMOS	30	$(1.4 \pm 0.3) \times 10^{-5}$	$(2.2 \pm 0.4) \times 10^{-5}$	-0.22 ± 0.1	1.2 ± 0.6
Boötes	90	$(2.3 \pm 0.6) \times 10^{-5}$	$(2.1 \pm 0.6) \times 10^{-5}$	0.05 ± 0.2	1.3 ± 0.7
COSMOS+ Boötes	120	$(2.1 \pm 0.3) \times 10^{-5}$	$(2.2 \pm 0.3) \times 10^{-5}$	-0.02 ± 0.1	1.2 ± 0.6

 Table 4.
 X-ray stacking results for the Chandra-undetected obscured IR AGN candidates in the COSMOS and Boötes fields.

Notes. Column description: (1) sample size; (2) mean count rate in the soft (0.5-2 keV) X-ray band; (3) mean count rate in the hard (2-7 keV) X-ray band; (4) the hardness radio H = (H - S)/(H + S) of the sample; and (5) median and standard deviation of the redshift distribution of the sample.



Figure 10. HR against redshift for the X-ray-undetected IR AGN candidates in the Boötes field (red), the COSMOS field (green), and the Boötes and COSMOS field combined (black). The X-ray stacking results are reported in Table 4. The dotted black and continuous lines represent the change of the HR with redshift for $N_{\rm H} = 10^{22}$ and 10^{23} cm⁻², respectively. The X-ray stacking results suggest that, on average, the X-ray-undetected source are obscured by $N_{\rm H} \gtrsim 10^{22}$ cm⁻².

the results in Section 3.2. In addition, the clear X-ray signal detected in the hard band strongly suggests that the majority of the Xray-undetected systems are real AGN, as we would not expect to detect a strong hard X-ray signal coming from inactive star-forming galaxies. Indeed, adopting the relationship between the SFR and 2 - 10 keV luminosity from Ranalli, Comastri & Setti (2003), the X-ray-undetected obscured IR AGN candidates would need to have, on average, SFR $\approx 750 \text{ M}_{\odot} \text{ yr}^{-1}$ to produce the stacked X-ray counts observed in the hard X-ray band. However, our SED fitting finds that in the COSMOS and Bootes fields, the mean SFR of the X-rayundetected obscured IR AGN candidates is SFR $\approx 25^{+181}_{-22} M_{\odot} \text{ yr}^{-1}$ (which is consistent with the SFR of all obscured IR AGN candidates; see next subsection), and that only $\sim 4 \text{ per cent}(4/110)$ of the Xray-undetected sources have SFR > 750 M_{\odot} yr⁻¹. Therefore, we are confident that the strong stacked hard X-ray signal is dominated by AGN processes, and only has a small contribution from SF. Moreover, the population could include many CT AGN with intrinsically hard but faint observed soft X-ray emission due to scattering (e.g. Carroll et al. 2021, 2023).

Assuming the same fraction of X-ray-undetected systems are obscured by $N_{\rm H} > 10^{22}$ cm⁻² as found for the X-ray-detected obscured IR AGN candidates (i.e. ~ 70 per cent), we predict that the 4MOST IR AGN survey will select ~ 148 000 X-ray obscured IR AGN with $N_{\rm H} > 10^{22}$ cm⁻². Table 5 summarizes the predictions made in this section for the AGN luminosity and obscuration properties of the final 4MOST IR AGN surveys.

4.2.2 Host galaxy properties: SFR and stellar masses

Our SED fitting also provides constraints on the $8 - 1000 \,\mu\text{m}$ SF luminosity $L_{SE,IR}$ and stellar masses of the sample, allowing us to make predictions for the host-galaxy properties of the 4MOST IR AGN sample. We estimate the star-formation rates (SFRs) using $L_{\text{SF,IR}}$ and assuming the Kennicutt (1998) relation. Fig. 11 shows the SFR against the stellar masses for the obscured IR AGN candidates down to $r_{AB} < 22.1$ and < 22.8. Overall, we find that the host galaxy properties of the wide and deep samples do not significantly differ. Our SED fitting could reliably measure the SFR for ~ 45 per cent of the sample, while for the other ~ 55 per cent, we can only provide an upper limit, either because of a lack of sufficient photometric coverage or because the AGN emission dominates the IR SED. For sources with $r_{AB} < 22.1$, the SFR ranges $\approx 4 - 1300 \text{ M}_{\odot} \text{ yr}^{-1}$, with a median value of $\approx 27^{+172}_{-22} \text{ M}_{\odot} \text{ yr}^{-1}$, and for sources with $r_{AB} <$ 22.8, the SFR ranges $\approx 2-3300~M_{\odot}~yr^{-1},$ with a median value of \approx 27^{+194}_{-23} M_{\odot} yr⁻¹. Although the median SFR values of the wide (r_{AB} < 22.1) and deep ($r_{AB} < 22.8$) samples are consistent, we find that the *p*-value of the SFR distributions is ≈ 0.1 , which provides marginal evidence for a statistical difference between both distributions. This is probably due to the fact that the deep sample has more sources with large SFRs (which is reflected in its modestly larger SFR standard deviation), which is expected given that the deep sample also has higher redshifts. These values remain almost the same when we only consider the sources that are also identified as IR AGN by the SED fitting.

We also find that the obscured IR AGN candidate host galaxies are generally massive, with stellar masses ranging $\log M_{\star}/M_{\odot} \approx$ 9 – 12.5, and with a median value of $M_{\star}/M_{\odot} \approx$ 10.8 ± 0.6 for both the wide and deep samples, respectively. In addition, we find that the *p*-value between the deep and wide samples is \approx 0.45, indicating that there are no significant differences in the stellar masses of both samples.

We also calculate the host galaxy properties of the IR quasars with $L_{\rm AGN,IR} > 10^{45} {\rm ~erg~s}^{-1}$ only. We find that the IR quasars are hosted in more massive $(M_{\star}/M_{\odot} \approx 11.1 \pm 0.5)$ and more powerful star-forming (SFR $\approx 50^{+270}_{-43} {\rm ~M}_{\odot} {\rm yr}^{-1})$ host galaxies than the overall IR AGN sample, consistent with these more luminous AGN hosting more massive SMBHs. The SFR and stellar masses results are reported in Table 5.

Motivated by the results of Andonie et al. (2022, 2024), who found a large fraction of IR quasars residing in starburst host galaxies,

Table 5. Predictions for some AGN and host galaxy properties of all the obscured IR AGN candidates in the deep ($r_{AB} < 22.8$) and wide ($r_{AB} < 22.1$) components of the 4MOST IR AGN survey, based on the UV-to-FIR SED-fitting and X-ray analyses in the deep fields.

4MOST sample	Ν	Z	N _{IR quasar}	$N_{N_{ m H}>10^{22}}$	$\log L_{\rm AGN, IR}$ (erg s ⁻¹)	$\log M_{\star}$ (M _☉)	$\log L_{\rm SF,IR}$ (erg s ⁻¹)	$\frac{\text{SFR}}{(\text{M}_{\odot} \text{ yr}^{-1})}$
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
IR AGN wide IR AGN deep	$ \begin{array}{c} \sim 182000 \\ \sim 30000 \end{array} $	$\begin{array}{c} 0.8\pm0.6\\ 1\pm0.6 \end{array}$	$ \begin{array}{l} \sim 100000 \\ \sim 16500 \end{array} $	$\begin{array}{c} \sim 127000 \\ \sim 21000 \end{array}$	$\begin{array}{c} 45.1 \pm 0.8 (45.6 \pm 0.5) \\ 45 \pm 0.9 (45.6 \pm 0.5) \end{array}$	$\begin{array}{c} 10.8 \pm 0.6 \ (11.1 \pm 0.5) \\ 10.8 \pm 0.6 \ (11.1 \pm 0.5) \end{array}$	$\begin{array}{c} 44.8 \pm 0.8 \ (45 \pm 0.8) \\ 44.8 \pm 0.9 \ (45 \pm 0.8) \end{array}$	$\begin{array}{c} 27^{+172}_{-22} \ (50^{+270}_{-42}) \\ 27^{+194}_{-23} \ (50^{+270}_{-42}) \end{array}$

Notes. Column description: (1) the sample; (2) sample size; (3) median and standard deviation of the redshift; (4) predicted number of IR quasars with $L_{AGN,IR} > 10^{45}$ erg s⁻¹; (5) predicted number of X-ray obscured AGN with $N_{\rm H} > 10^{22}$ cm⁻² (number based on both X-ray-detected AGN and stacking analyses of X-ray-undetected AGN); (6) median and standard deviation of the 8 – 1000 μ m AGN luminosity ($L_{AGN,IR}$) of the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields; (7) median and standard deviation of the stellar masses for the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields; (8) median and standard deviation of the 8 – 1000 μ m SF luminosity ($L_{\rm SF,IR}$) of the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields; and (9) median and standard deviation of the SFRs calculated from $L_{\rm SF,IR}$ for the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields; and (9) median and standard deviation of the SFRs calculated from $L_{\rm SF,IR}$ for the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields; and (9) median and standard deviation of the SFRs calculated from $L_{\rm SF,IR}$ for the obscured IR AGN candidates and the IR quasars (in parenthesis) in the deep fields.



Figure 11. SFR against the stellar mass M_{\star} of the obscured IR AGN candidates in the deep fields down to $r_{AB} < 22.1$ mag (black/grey) and $r_{AB} < 22.8$ mag (red/light red) as a representation of the 4MOST IR AGN wide and deep surveys, respectively. The SFRs are calculated from the 8 – 1000 μ m star formation luminosity. The cyan curves represent the main sequence of star-forming galaxies relation from Schreiber et al. (2015) for z = 0.5 (dotted curve), z = 1 (solid curve), and z = 1.5 (dashed curve), converted from the Salpeter to the Chabrier IMF (see Footnote 4.2.2). The top histograms show the distribution of M_{\star} while the right histograms show distributions of SFR. In both cases, the filled light-colour histograms represent the well-constrained SFR and M_{\star} , while the dotted, empty histograms represent the 99th percentile of the posterior distribution of unconstrained SFR and M_{\star} .

we calculated the amount of obscured IR AGN candidates in the starburst phase adopting the main sequence of star-forming galaxies relation from Schreiber et al. $(2015)^{21}$ (see the cyan curves in

Fig. 11). We classify a source as in the starburst phase when it is > 0.3 dex above the SFR- M_{\star} main sequence relationship. We find 123 and 198 IR AGN candidates residing in starburst host galaxies down to $r_{AB} < 22.1$ and < 22.8, respectively, which corresponds to ≈ 33 per cent of each sample. This fraction is in full agreement with Andonie et al. (2022), who also found that ~ 29 per cent (170/578) of their SED-selected IR quasar sample in the starburst phase. Extrapolating to the rest of 4MOST IR AGN survey footprint, we predict that the 4MOST IR AGN survey will identify $\sim 70\,000$

²¹We rescale the Schreiber et al.'s (2015) SFR- M_{\star} relationship from the Salpeter (Salpeter 1955) to the Chabrier IMF adopting the following commonly used relations reported in Madau & Dickinson (2014): SFR_{Chabrier} = $0.64 \times SFR_{Salpeter}$ and $M_{\star,Chabrier} = 0.61 \times M_{\star,Salpeter}$.



Figure 12. Left panel: r - W2 against redshift of the *eROSITA* AGN in the eFEDS (eFEDS X-ray AGN, grey dots), equivalent to the 4MOST X-ray AGN wide survey, and obscured IR AGN candidates in the deep fields down to $r_{AB} = 22.8$ mag (green stars, see Section 4.2). The contours show where the samples lie at 50 per cent, 68 per cent, and 90 per cent confidence levels. The figure also shows the distribution of the redshifts (top histograms) and r - W2 (right histograms) for the eFEDS X-ray AGN (grey dashed lines) and obscured IR AGN candidates (green solid lines). Right panel: same as left panel, but only showing the eFEDS X-ray AGN that are also identified as IR AGN by the A18 R90 selection, using unWISE photometry (eFEDS X-ray and IR AGN, blue dots and blue dotted lines).

IR AGN residing in starburst host galaxies, where the SF is likely to provide a significant (potentially dominant) contribution to the AGN obscuration (e.g. Circosta et al. 2019; Gilli et al. 2022; Andonie et al. 2024).

4.3 Comparison to the 4MOST X-ray AGN survey with *eROSITA*

The 4MOST X-ray AGN survey will target ~ 1 M AGN selected by eROSITA. The 4MOST IR AGN survey is designed to complement the 4MOST X-ray AGN survey by identifying the obscured AGN that eROSITA will miss due to its predominantly soft X-ray sensitivity response. In this subsection, we use the eROSITA observations in eFEDS (hereafter eFEDS X-ray AGN) as an equivalent of the 4MOST X-ray AGN survey to demonstrate that our 4MOST IR AGN survey will target AGN with different properties to the 4MOST X-ray AGN sample. The eFEDs region is the ideal field for these analyses since the *eROSITA* observations are of a comparable depth (actually $\sim \times 2$ times deeper) to those utilized in the 4MOST Xray AGN survey and it has good-quality multiwavelength follow-up observations and optical spectroscopic redshifts.²² We note that we do not use the publicly available first six months of the eROSITA all-sky survey (eRASS1; Merloni et al. 2024) because it does not have redshift estimates and is much shallower than the eROSITA observations used to define the 4MOST X-ray AGN survey.

To compare the properties of the 4MOST IR AGN and the 4MOST X-ray AGN surveys, we would ideally compare the obscured

IR AGN candidates in the eFEDS footprint with the eFEDS Xray AGN. However, the majority of the eFEDS area does not have extensive optical spectroscopic redshifts of non-X-ray sources or multiwavelength photometry, and therefore lacks the accurate redshifts and SED-fitting constraints we require for the obscured IR AGN analyses. Instead, we decided to use the sources that meet our 4MOST obscured IR AGN selection in the deep fields (see Section 4.2) to perform the redshift comparison with the eFEDS X-ray AGN.

Fig. 12 depicts the r - W2 colour versus redshift for both samples, showing with contours where the samples lie at 50 per cent, 68 per cent ($\sim 1\sigma$), and 90 per cent confidence level. The left panel plots the eFEDS X-ray AGN with a W2 S/N> 3 in the LSDR8, equivalent to 21 320 sources, which clearly shows that the eFEDS X-ray AGN and the obscured IR AGN candidates have different r - W2 colours. By design, all the obscured IR AGN candidates have $r - W^2 > 5.9$ mag, while only ~ 21 per cent(4425/21320) of the eFEDS X-ray AGN have $r - W2 \ge 5.9$ mag. We perform a Kolmogorov–Smirnov test and we find a p – value $\ll 0.001$, confirming that both distributions are different. This plot also shows that the eFEDS X-ray AGN and the obscured IR AGN candidates have different redshift distributions (p - value $\ll 0.001$); the eFEDS X-ray AGN have more sources at $z \leq 0.5$ and $\gtrsim 1.5$. Indeed, we find that \sim 19 per cent and \sim 34 per cent of the eFEDS X-ray AGN have z < 0.5 and > 1.5, respectively, while only ~ 9 per cent and ~ 20 per cent of the obscured IR AGN candidates have z < 0.5 and > 1.5, respectively. These differences are likely due to the X-ray data identifying lower luminosity AGN at lower redshifts (where the host-galaxy will dilute the AGN signal in the IR waveband), and unobscured AGN at higher redshifts which are bright in the optical waveband due to a direct view of the nucleus.

Fig. 13 depicts the *r*-band distributions of the eFEDS X-ray AGN and the obscured IR AGN candidates, showing that they also differ (p – value \ll 0.001). The eFEDS X-ray AGN have an *r*-band distribution which resembles a Gaussian centred around $r_{AB} \approx 21.1$;

²²We note that the eFEDS has much shallower X-ray observations than the *Chandra* and *XMM–Newton* data that we used in the deep field analyses, and does not have MIR and FIR photometric coverage. Hence, we cannot perform the same detailed UV-to-FIR SED-fitting and X-ray analyses that we performed in the deep fields.



Figure 13. Distributions of r_{AB} of the *eROSITA* AGN in the eFEDS (black dashed lines), the *eROSITA* AGN in the eFEDS that are also identified as IR AGN by the A18 R90 selection using unWISE photometry (blue dotted lines), and the obscured IR AGN candidates in the deep fields down to $r_{AB} = 22.8 \text{ mag}$ (green continuous line).

i.e. half of the sources have $r_{AB} < 21.1$. On the other hand, the obscured IR AGN have an *r*-band distribution which resembles an inverse exponential function, where the *r* band rapidly rises from $r_{AB} \approx 21$ to a peak at the limit of our 4MOST IR AGN survey. Indeed, we find, that ~ 80 per cent of the obscured IR AGN have $r_{AB} > 21.1$. The fainter optical magnitudes for the obscured IR AGN are expected since the obscuration will extinguish most of the bright accretion disc emission in the optical band (see Fig. 5c). We note that the eFEDS X-ray AGN that are also IR AGN are optically brighter than the overall eFEDS X-ray AGN population. This apparently surprising result is likely due to the soft X-ray sensitivity of *eROSITA* meaning that only the brightest obscured AGN can be detected, demonstrating that *eROSITA* only selects an extreme end of the obscured AGN population.

To provide a more direct prediction of the *eROSITA* properties we also investigate the 4MOST IR AGN survey sources in the eFEDS footprint: overall, \sim 4476 and \sim 2594 obscured IR AGN candidates reside in eFEDs with $r_{AB} < 22.8$ and < 22.1, respectively. We find that only ~ 18 per cent (856 out of 4476) and ~ 24 per cent (633 out of 2594) 4MOST IR AGN sources are detected by eROSITA, down to $r_{AB} < 22.8$ and < 22.1, respectively; see Table 6. The remaining $\sim 76-82$ per cent are undetected, probably due to the soft X-ray emission being absorbed by large column densities, as predicted by our analyses in Section 4.2.1. We can directly test this scenario using the X-ray spectral constraints in eFEDS from Liu et al. (2022). Interestingly, we find that the eROSITA-detected 4MOST IR AGNs have a median log $N_{\rm H}/({\rm cm}^{-2}) \approx 21.2 {\rm cm}^{-2}$, while the *eROSITA* sample in eFEDs have a median log $N_{\rm H}/({\rm cm}^{-2}) \approx 20.8 {\rm ~cm}^{-2}$. Additionally, the obscured ($N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$) fraction of the *eROSITA*-detected 4MOST IR AGNs is ~ 22 per cent, which is ~ $3 \times$ larger than for the entire eFEDS sample (Liu et al. 2022). We note that the eROSITA-detected 4MOST IR AGNs are significantly less obscured than the obscured IR AGN candidates in Section 4.2.1 and the X-ray AGN in Section 3.2 because the eROSITA observations in eFEDs are significantly shallower than Chandra and XMM-Newton in the deep fields, missing the most obscured AGNs. Similarly, Toba et al. (2022) analysed the properties of WISE 22 μ m (W4) detected IR AGN

candidates in eFEDS and found that only ~ 10 per cent are detected by *eROSITA*. They concluded, in agreement with our results here, that the wide majority of the *WISE* 22 μ m sources are undetected by *eROSITA* due to extreme absorption.

Finally, we apply a self-consistent IR AGN selection to the eFEDS eROSITA X-ray AGN, identifying the A18 R90 IR AGN using the unWISE magnitudes.²³ This analysis will inform us of the total sample of IR AGNs that will be targeted by the 4MOST X-ray and IR AGN surveys combined. The expected r - W2 and redshift distributions are shown in the right panel of Fig. 12. We find that ~ 26 per cent (5785 out of 22 079) and ~ 25 per cent (5574 out of 22079) of the eFEDS X-ray AGN meet the A18 R90 IR AGN selection down to $r_{AB} < 22.8$ and < 22.1, respectively. The numbers are almost the same for $r_{AB} < 22.8$ and < 22.1, since most eFEDS X-ray AGN are optically bright. We find that the majority (~ 86 per cent: 5001 out of 5785) of the IR-eFEDS X-ray AGN have r - W2 < 5.9 mag colours, as expected given the majority of the eROSITA sources are X-ray unobscured. Extrapolating to the rest of the 4MOST IR AGN footprint, we expect that the 4MOST X-ray AGN survey will identify \approx 300 000 IR AGNs with r - W2 < 5.9 mag colours, directly complementing our $\approx 200\,000$ obscured IR AGN with $r - W2 \ge 5.9$ mag. Hence, the combination of the 4MOST IR AGN and the 4MOST X-ray AGN surveys will produce a sample of $\sim 500\,000$ IR-selected AGN, the largest optical spectroscopic survey targeting IR AGNs to date by > 2 orders of magnitude. Table 6 summarizes the statistics reported in this section.

4.4 Scientific goals

The 4MOST IR AGN survey will provide the first large-scale optical spectroscopic sample of IR AGN to investigate the accretion and host galaxy properties of obscured IR AGN. When combined with the IR-detected sources in the 4MOST X-ray AGN survey, it will provide the most comprehensive and complete census of the IR AGN population to date, allowing us to determine what (sub) populations have been missed in previous studies, test whether obscured IR AGN are fundamentally different to unobscured IR AGN, and to understand whether obscured IR AGN preferentially reside in different environments to unobscured IR AGN. This will be achieved from the combination of the 4MOST optical spectra along with the rich multiwavelength coverage overlapping the 4MOST IR AGN survey footprint (see Fig. 2). Here, we describe some of the specific analyses that can be performed with this comprehensive data set to characterize the AGN and host galaxy properties of the IR AGN population.

(i) Spectroscopic characterization. The 4MOST optical spectra will allow, for the first time, the large-scale exploration of the optical spectral properties of the obscured IR AGN population. We will use the spectra to measure redshifts and constrain the accretion properties of our sources. Due to the 4MOST spectral coverage (3700 – 9000 Å), we expect to detect many important AGN emission lines such as [Ne V] λ 3346, 3426 Å, [He II] λ 4687 Å, and [O III] λ 5008 Å; see Fig. 7. These AGN emission lines will probe AGN activity up to $z \sim 1.5$, where most of the 4MOST IR AGN sources. For example, we can use different emission-line diagnostic to identify optical AGN signatures in our sample such as 'BPT'

 $^{^{23}}$ To find the unWISE counterparts of the eFEDS sources, we match the optical positions of the *eROSITA* AGN in eFEDs with the unWISE catalogue, adopting the same approach as that outlined in Section 2.2

Table 6. Sample statistics for the *eROSITA* AGN and the 4MOST IR AGN deep ($r_{AB} < 22.8$) and 4MOST IR AGN wide ($r_{AB} < 22.1$) surveys in the eFEDS footprint.

Sample (1)	N (2)	Redshift (3)	<i>r</i> _{AB} (4)	f _{eRO} (5)	$f_{N_{\rm H}>10^{22}}$ (6)
eFEDS X-ray AGN eFEDS X-ray and IR AGN	22 079 5912	$\begin{array}{c} 1.2\pm0.8\\ 1.3\pm0.9\end{array}$	21.1 ± 1.7 19.9 ± 1.2	1 1	~ 0.08 ~ 0.07
4MOST IR AGN deep 4MOST IR AGN wide	4676 2594	$\begin{array}{c} 0.8\pm0.6\\ 1\pm0.6 \end{array}$	$\begin{array}{c} 22\pm0.6\\ 21.6\pm0.5\end{array}$	0.18 (856/4676) 0.24 (633/2594)	~ 0.68 ~ 0.68

NoItes. Column descriptions: (1) the sample; (2) sample size; and (3) median and standard deviation of the redshift. For the 4MOST samples, the redshift statistics are calculated using the predictions in the deep fields (see Section 4.2.1); (4) median and standard deviation of the *r*-band values in AB units; (5) the fraction of the sample detected by *eROSITA* in eFEDS; and (6) the fraction of the sample expected to have $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$.

diagnostics out to $z \sim 0.35$ (e.g. Baldwin, Phillips & Terlevich 1981; Kauffmann et al. 2003), or $[O III]/H\beta$ and [Ne III]/[O II] versus [OIII]/[OII] emission-line ratio analyses at higher redshifts (e.g. Yan et al. 2011; Juneau et al. 2011; Feuillet et al. 2024). The optical spectra will also facilitate other analyses, depending on the detected emission lines and S/N (e.g. energetic outflows from broadened forbidden lines; black-hole masses from broad permitted lines; dust obscuration from continuum and emission-line fitting; e.g. McLure & Jarvis 2002; Vestergaard & Peterson 2006; Shen et al. 2011; Fawcett et al. 2022; Rakshit et al. 2020). In the case of the systems without clear optical, X-ray, or IR SED AGN signatures but reliable redshift, we can use optical spectral stacking to study different subsets of the IR AGN population (e.g. extremely red quasars and hot dust obscured galaxies; e.g. Stern et al. 2014; Assef et al. 2015; Ross et al. 2015; Hamann et al. 2017) and search for faint AGN features (e.g. Zhu et al. 2015; Rigby et al. 2018; Arnaudova et al. 2024). Little is known about the overall obscured IR AGN population and it is also quite probable that we will discover new subpopulations of obscured AGN.

(ii) *Dust-obscured SFR*. Rhe 4MOST footprint overlaps with ten well-observed fields of the sky with *Herschel* coverage, which together cover an area of \sim 480 deg²; see Fig. 2. In those fields, we can use UV-to-FIR SED fitting to reliably calculate the SFR of the IR AGN sample. Combined with the stellar masses, we can also explore where the obscured IR AGN lie on the main sequence of SF (e.g. Noeske et al. 2007; Whitaker et al. 2012; Schreiber et al. 2015), and investigate the incidence and properties of extreme starbursts in IR quasars, extending the results found by Andonie et al. (2022, 2024).

(iii) X-ray analyses. For the ~ 20 per cent of the obscured IR AGN that will be detected by *eROSITA*, we can perform X-ray spectral fitting to constrain key AGN properties such as X-ray luminosity and $N_{\rm H}$ to compare with the 4MOST X-ray AGN survey. In addition, we can perform more detailed X-ray spectral analyses in the fields with deeper X-ray data from *Chandra* and *XMM–Newton* (which typically align with the fields also with *Herschel* FIR coverage). For the ~ 80 per cent of the sample that will be X-ray undetected, we can use a similar approach to that adopted in Section 4.2.1, and perform X-ray stacking analyses to assess the level of obscuration for different AGN subpopulations (e.g. Carroll et al. 2021; Andonie et al. 2022; Carroll et al. 2023).

(iv) SED analysis. All our sources have information in the optical bands g, r, i, and z, and the WISE bands W1 and W2. We also find that ~ 60 per cent and ~ 40 per cent of our sources have $> 2\sigma$ detections in W3 and W4, respectively. This combination of photometry already allows us to characterize the UV-to-MIR SED of the entire 4MOST IR AGN survey. In addition, the X-ray analysis will provide the X-ray

luminosity and obscuring gas column ($N_{\rm H}$) for individual sources, which can be used to place priors and improve the constraints on the contribution of the AD emission to the rest-UV SED, as adopted in Section 2.3.2. With this information, we can perform a simple UV-to-MIR SED fitting, and retrieve the AGN luminosity and stellar masses of the entire sample and make comparisons between different subpopulations. To demonstrate the viability of this analysis, we have developed an SED-fitting approach that uses the physical constraints from the deep-field analysis (see Section 4.2) as priors to help guide the SED fitting in the rest of the sky with less photometric coverage. We find that this approach will be particularly useful to constrain the average AGN luminosity and stellar masses of AGN samples, but the values of individual sources are more uncertain. Appendix B describes this approach and shows its performance.

(v) *Placing obscured IR AGN within the overall AGN population.* The 4MOST IR and X-ray AGN surveys will provide a rich and large sample that can be used to compare the AGN, host galaxy properties, and large-scale environments of IR and X-ray-selected AGN as a function of different physical parameters to determine the relationship between obscured and unobscured AGN and test the standard AGN orientation model. These comparisons can be used to study the incidence of obscuration in AGN, the distribution of the AGN obscuring material (e.g. Merloni et al. 2014), the galaxy clustering of different AGN populations (e.g. Petter et al. 2022, 2023), and to test SMBH-galaxy co-evolution scenarios (Andonie et al. 2022, 2024).

5 SUMMARY

We present the motivation and design of the 4MOST IR AGN survey, the first large-scale optical spectroscopic survey targeting MIRselected obscured AGN. Our parent samples are the LS DR10 and the unWISE catalogues. To select the obscured AGN, we first identify the IR AGN using the Assef et al. (2018) 90 per cent AGN reliability colour selection using unWISE magnitudes, and then identify the obscured IR AGN selecting the objects with $r - W2 \ge 5.9$ mag. The final 4MOST IR AGN survey will target $\approx 212\,000$ sources and has two components: a deep component which will target $\approx 30\,000$ sources down to $r_{AB} < 22.8$ over a well-observed region of $\sim 850 \text{ deg}^2$, and a wide component which will target $\approx 182\,000$ sources down to $r_{AB} < 22.1$ over a larger area of $\sim 9000 \text{ deg}^2$. Our survey aims to complement the 4MOST X-ray AGN survey, which will target $\sim 1M \text{ eROSITA}$ AGN, by identifying the obscured AGN that eROSITA will miss across the same spatial footprint.

To investigate the expected properties of the 4MOST IR AGN survey selection function, we analyse a sample of ~ 650 obscured IR AGN candidates in four deep, well-observed fields in the sky

using detailed UV-to-FIR SED-fitting and X-ray spectral constraints from *Chandra* or *XMM–Newton*. The deep fields analysis provides information on the AGN reliability of our adopted approach and helps to predict the AGN and host galaxy properties of the final 4MOST IR AGN survey sample. Our findings can be summarized as follows:

(i) Based on UV-to-FIR SED-fitting constraints, we find that the AGN reliability of our obscured IR AGN selection approach is ~ 80 per cent and ~ 87 per cent for $r_{AB} < 22.8$ and < 22.1, respectively. Our analysis finds that the AGN reliability of the A18 R90 when applied to unWISE photometry is ~ 70 per cent; however, the majority of the contamination is coming from sources with r - W2 < 4 mag. Hence, after removing those sources, the reliability of the Assef et al. (2018) selection increases to ~ 80 - 90 per cent (see Section 3.3 and Appendix A).

(ii) We find that the optical-to-MIR colour cut $r - [4.5] \ge 5.9$ mag, adapted from Hickox et al. (2007), is a good proxy of AGN obscuration. The Hickox et al. (2007) selection was defined using the *Spitzer* IRAC 2 channel at 4.5 μ m, but we have demonstrated it can also be used with *WISE* W2 at 4.6 μ m. We tested the colour cut using X-ray spectral constraints in the COSMOS field, and find that $N_{\rm H}$ systematically increases with r - [4.5]/W2 (see Fig. 5). We find that sources with $r - [4.5]/W2 \ge 5.9$ mag have a median $N_{\rm H,r-4.5/W2 \ge 5.9} = 3.5 \times 10^{22}$ cm⁻², and ~ 65 per cent of them have a $N_{\rm H} > 10^{22}$ cm⁻². On the other hand, sources with r - [4.5]/W2 < 5.9 mag have a median $N_{\rm H,r-4.5/W2 < 5.9} = 3 \times 10^{21}$ cm⁻² and only ~ 30 – 35 per cent of them have a $N_{\rm H} > 10^{22}$ cm⁻².

(iii) Based on the results of our optical spectral simulations, we predict that the observations from our survey will allow us to successfully calculate the redshifts (the overall recovery rate is ≈ 72 per cent but it increases to ≈ 82 per cent when excluding sources in the 'redshift desert') and identify prominent AGN emission lines in our targets. We find that the average exposure times of the 4MOST IR AGNs targets are ~ 14.5 and ~ 29.4 min for the wide and deep surveys, respectively; see Section 4.1.

(iv) On the basis of the UV-to-FIR SED-fitting results and X-ray analyses in the deep fields, we predict the AGN and host galaxy properties of the 4MOST IR AGN wide ($r_{AB} < 22.1$) and deep ($r_{AB} < 22.8$) surveys. We find that

(a) The obscuration properties of the obscured IR AGN candidates of the wide and deep surveys are similar with $\sim 45-50$ per cent detected by *Chandra* or *XMM–Newton*. The X

-ray constraints indicate that ~ 70 per cent of the X-raydetected sources have $N_{\rm H} > 10^{22} {\rm cm}^{-2}$. In addition, based on X-ray stacking analyses of the X-ray-undetected obscured IR AGN candidates, we conclude that the majority of the X-rayundetected sources are also obscured by $N_{\rm H} > 10^{22} {\rm cm}^{-2}$ (see Section 4.2.1).

(b) Our sources have redshifts ranging $\approx 0.5 - 3.5$, with median values of $z \approx 0.8$ and ≈ 1 for the wide and deep surveys, respectively. In addition, the AGN luminosities of both samples are consistent, with median values of log $L_{\text{AGN,IR}}/(\text{erg s}^{-1}) \approx 45.1$, where ≈ 55 per cent of sample are powerful IR quasars with $L_{\text{AGN,IR}} > 10^{45}$ erg s⁻¹ (see Section 4.2.1).

(c) The host galaxy properties of obscured IR AGN candidates are similar for the wide and deep surveys. Our sources are hosted by massive galaxies with a median stellar mass of log $M_{\star}/M_{\odot} \approx 10.8$. The SF properties of the obscured IR AGN candidates span a wide range, with SFR ranging $\approx 2 - 3300 M_{\odot} \text{ yr}^{-1}$, and a median value of $\sim 27 M_{\odot} \text{ yr}^{-1}$. In addition, we find that ≈ 33 per cent of the obscured IR AGN

candidates are hosted by galaxies lying within the starburst region above the main sequence of star-forming galaxies (see Section 4.2.2).

(v) We compare the properties of the 4MOST IR AGN and 4MOST X-ray AGN surveys in the eFEDS area, using the eROSITA observations in eFEDS as an equivalent to the 4MOST X-ray AGN survey (see Section 4.3). We find that ~ 17 per cent and \sim 22 per cent of the obscured IR AGN are detected by *eROSITA* in eFEDS down to $r_{AB} = 22.8$ and 22.1, respectively. We show that both surveys are complementary and target samples with different optical, MIR, and X-ray properties. We find (1) only ~ 21 per cent of the eFEDS X-ray AGN meet the obscured 4MOST IR AGN selection of $r - W^2 > 5.9$ mag; (2) the eFEDS X-ray AGN are optically bright with ~ 50 per cent of them having a $r_{AB} \lesssim 21.1$, while the 4MOST IR AGN sources are optically faint with ~ 80 per cent of them having $r_{AB} > 21.1$; (3) the 4MOST X-ray AGN survey covers a wider redshift range than the IR AGN survey with ~ 2.1 and ~ 1.7 times more sources at z < 0.5 and > 1.5, respectively, and (4) the majority (\sim 92 per cent) of the eFEDS X-ray AGN are X-ray unobscured with $N_{\rm H} < 10^{22} \,{\rm cm}^{-2}$, while the majority (~ 70 per cent) of the 4MOST IR AGN are expected to be obscured by $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}$.

On the basis of our analysis, we show that the 4MOST IR AGN survey will target a key population of $\sim 212\,000$ IR AGN, where ~ 140 000 sources will be obscured by $N_{\rm H} > 10^{22} \,{\rm cm}^{-2}, \sim 121\,000$ sources will be powerful IR quasars with $L_{AGN,IR} > 10^{45}$ erg s⁻¹, and only $\sim 40\,000$ sources will be detected by *eROSITA*. In addition, we predict that the 4MOST X-ray AGN survey will identify $\sim 300\,000$ IR AGN with r - W2 < 5.9 mag colours giving, when combined with the 4MOST IR AGN survey, $\sim 500\,000$ IR-selected AGN, which is $\sim 100 \times$ larger than any existing spectroscopic MIRselected AGN sample allowing for the first large-scale spectroscopic studies of this previously poorly studied population. Using the X-ray, optical/UV, and IR data overlapping with the 4MOST footprint, we will reliably characterize (both in the physical and empirical properties) the obscured IR AGN population for the first time and identify previously missed AGN subpopulations, extending our taxonomy of the overall AGN and quasar population. This rich data set will allow us to test whether (and how) obscured IR AGN are fundamentally different to unobscured IR AGN, distinguishing between the standard orientation-dependent unified model and evolutionary-based models, understand whether these differences are universal or restricted to key subpopulations, and to determine whether obscured IR AGN preferentially reside in different environments to unobscured IR AGN (see Section 4.4).

ACKNOWLEDGEMENTS

We thank the referee for their positive and constructive comments, which substantially improved the paper. This work has been supported by the EU H2020-MSCA-ITN-2019 Project 860744 'BiD4BESt: Big Data applications for black hole Evolution STudies'. CA is supported by the Alexander von Humboldt Foundation through a Humboldt Research Fellowship for postdoctoral researchers. DMA and CG thank the Science Technology Facilities Council (STFC) for support from the Durham consolidated grant (ST/T000244/1). DJR acknowledges the support of STFC grant NU-012097.

This research used data obtained with the Dark Energy Spectroscopic Instrument (DESI). DESI construction and operations is managed by the Lawrence Berkeley National Laboratory. This material is based upon work supported by the U.S. Department of Energy, Office of Science, Office of High-Energy Physics, under contract no. DE-AC02-05CH11231, and by the National Energy Research Scientific Computing Center, a DOE Office of Science User Facility under the same contract. Additional support for DESI was provided by the U.S. National Science Foundation (NSF), Division of Astronomical Sciences under contract no. AST-0950945 to the NSF's National Optical-Infrared Astronomy Research Laboratory; the Science and Technology Facilities Council of the United Kingdom; the Gordon and Betty Moore Foundation; the Heising-Simons Foundation; the French Alternative Energies and Atomic Energy Commission (CEA); the National Council of Science and Technology of Mexico (CONACYT); the Ministry of Science and Innovation of Spain (MICINN), and by the DESI Member Institutions: www.desi.lbl.gov/collaborating-institutions. The DESI collaboration is honoured to be permitted to conduct scientific research on Iolkam Du'ag (Kitt Peak), a mountain with particular significance to the Tohono O'odham Nation. Any opinions, findings, and conclusions or recommendations expressed in this material are those of the author(s) and do not necessarily reflect the views of the U.S. National Science Foundation, the U.S. Department of Energy, or any of the listed funding agencies.

DATA AVAILABILITY

The data sets generated and/or analysed in this study are available from the corresponding author on reasonable request.

REFERENCES

- Aird J., Coil A. L., Georgakakis A., Nandra K., Barro G., Pérez-González P. G., 2015, MNRAS, 451, 1892 Alexander D. M. et al., 2003, AJ, 126, 539 Alexander D. M. et al., 2008, ApJ, 687, 835
- Alexander D. M. et al., 2023, AJ, 165, 124
- Alexander D. M., Hickox R. C., 2012, New Astron. Rev., 56, 93
- Alexander D. M., Vignali C., Bauer F. E., Brandt W. N., Hornschemeier A. E., Garmire G. P., Schneider D. P., 2002, AJ, 123, 1149
- Ananna T. T. et al., 2019, ApJ, 871, 240
- Ananna T. T., Bogdán Á., Kovács O. E., Natarajan P., Hickox R. C., 2024, ApJ, 969, L18
- Andonie C. et al., 2022, MNRAS, 517, 2577
- Andonie C. et al., 2024, MNRAS, 527, L144
- Arnaudova M. I. et al., 2024, MNRAS, 528, 4547
- Assef R. J. et al., 2013, ApJ, 772, 26
- Assef R. J. et al., 2015, ApJ, 804, 27
- Assef R. J., Stern D., Noirot G., Jun H. D., Cutri R. M., Eisenhardt P. R. M., 2018, ApJS, 234, 23 (A18)
- Baldwin J. A., Phillips M. M., Terlevich R., 1981, PASP, 93, 5
- Baronchelli L., Nandra K., Buchner J., 2020, MNRAS, 498, 5284
- Barrows R. S., Comerford J. M., Stern D., Assef R. J., 2021, ApJ, 922, 179
- Bauer F. E. et al., 2023, The Messenger, 190, 34
- Baumgartner W. H., Tueller J., Markwardt C. B., Skinner G. K., Barthelmy S., Mushotzky R. F., Evans P. A., Gehrels N., 2013, ApJS, 207, 19
- Blecha L., Snyder G. F., Satyapal S., Ellison S. L., 2018, MNRAS, 478, 3056
- Brandt W. N. et al., 2018, preprint (arXiv:1811.06542)
- Brandt W. N., Alexander D. M., 2015, A&AR, 23, 1
- Brunner H. et al., 2022, A&A, 661, A1 Brusa M. et al., 2005, A&A, 432, 69
- Brusa M. et al., 2010, ApJ, 716, 348
- Brusa M., Comastri A., Daddi E., Cimatti A., Mignoli M., Pozzetti L., 2002, ApJ, 581, L89
- Bruzual G., Charlot S., 2003, MNRAS, 344, 1000
- Buchner J. et al., 2014, A&A, 564, A125
- Buchner J., Brightman M., Nandra K., Nikutta R., Bauer F. E., 2019, A&A, 629. A16
- Burtscher L. et al., 2016, A&A, 586, A28

- Calzetti D., Armus L., Bohlin R. C., Kinney A. L., Koornneef J., Storchi-Bergmann T., 2000, ApJ, 533, 682
- Carroll C. M., Ananna T. T., Hickox R. C., Masini A., Assef R. J., Stern D., Chen C.-T. J., Lanz L., 2023, ApJ, 950, 127
- Carroll C. M., Hickox R. C., Masini A., Lanz L., Assef R. J., Stern D., Chen C.-T. J., Ananna T. T., 2021, ApJ, 908, 185
- Chabrier G., 2003, ApJ, 586, L133
- Chaussidon E. et al., 2023, ApJ, 944, 107
- Chen C.-T. J. et al., 2015, ApJ, 802, 50
- Chiappetti L. et al., 2018, A&A, 620, A12
- Circosta C. et al., 2019, A&A, 623, A172
- Civano F. et al., 2016, ApJ, 819, 62
- Comastri A., Setti G., Zamorani G., Hasinger G., 1995, A&A, 296, 1
- Comparat J. et al., 2020, A&A, 636, A97
- Dale D. A., Helou G., Magdis G. E., Armus L., Díaz-Santos T., Shi Y., 2014, ApJ, 784, 83
- de Jong J. T. A., Verdoes Kleijn G. A., Kuijken K. H., Valentijn E. A., 2013, Exper. Astron., 35, 25
- de Jong R. S. et al., 2019, The Messenger, 175, 3
- Del Moro A. et al., 2016, MNRAS, 456, 2105
- DESI Collaboration, 2016, preprint (arXiv:1611.00036)
- DESI Collaboration, 2024, AJ, 168, 58
- Dey A. et al., 2019, AJ, 157, 168
- Di Matteo T., Springel V., Hernquist L., 2005, Nature, 433, 604
- DiPompeo M. A., Hickox R. C., Carroll C. M., Runnoe J. C., Mullaney J. R., Fischer T. C., 2018, ApJ, 856, 76
- DiPompeo M. A., Myers A. D., Hickox R. C., Geach J. E., Hainline K. N., 2014, MNRAS, 442, 3443
- DiPompeo M. A., Myers A. D., Hickox R. C., Geach J. E., Holder G., Hainline K. N., Hall S. W., 2015, MNRAS, 446, 3492
- Donley J. L. et al., 2012, ApJ, 748, 142
- Dougherty S. L., Harrison C. M., Kocevski D. D., Rosario D. J., 2024, MNRAS, 527, 3146
- Driver S. P. et al., 2019, The Messenger, 175, 46
- Duncan K. et al., 2023, The Messenger, 190, 25
- Eales S. et al., 2010, PASP, 122, 499
- Edge A., Sutherland W., Kuijken K., Driver S., McMahon R., Eales S., Emerson J. P., 2013, The Messenger, 154, 32
- Fawcett V. A. et al., 2023, MNRAS, 525, 5575
- Fawcett V. A., Alexander D. M., Rosario D. J., Klindt L., Lusso E., Morabito L. K., Calistro Rivera G., 2022, MNRAS, 513, 1254
- Feroz F., Hobson M. P., 2008, MNRAS, 384, 449
- Feroz F., Hobson M. P., Bridges M., 2009, MNRAS, 398, 1601
- Feroz F., Hobson M. P., Cameron E., Pettitt A. N., 2019, Open J. Astrophys., 2.10
- Ferrarese L., Merritt D., 2000, ApJ, 539, L9
- Feuillet L. M., Meléndez M., Kraemer S., Schmitt H. R., Fischer T. C., Reeves J. N., 2024, ApJ, 962, 104
- Fiore F. et al., 2008, ApJ, 672, 94
- Flaugher B. et al., 2015, AJ, 150, 150
- Gebhardt K. et al., 2000, ApJ, 539, L13
- Gilli R. et al., 2022, A&A, 666, A17
- Gilli R., Comastri A., Hasinger G., 2007, A&A, 463, 79
- Greenwell C. L. et al., 2024, ApJS, 273, 20
- Hamann F. et al., 2017, MNRAS, 464, 3431
- Hickox R. C. et al., 2007, ApJ, 671, 1365
- Hickox R. C. et al., 2011, ApJ, 731, 117
- Hickox R. C., Alexander D. M., 2018, ARA&A, 56, 625
- Hickox R. C., Myers A. D., Greene J. E., Hainline K. N., Zakamska N. L., DiPompeo M. A., 2017, ApJ, 849, 53
- Hinshaw G. et al., 2013, ApJS, 208, 19
- Hopkins P. F., Hernquist L., Cox T. J., Kereš D., 2008, ApJS, 175, 356
- Jin S. et al., 2018, ApJ, 864, 56
- Juneau S., Dickinson M., Alexander D. M., Salim S., 2011, ApJ, 736, 104
- Kauffmann G. et al., 2003, MNRAS, 346, 1055
- Kennicutt Robert C. J., 1998, ARA&A, 36, 189
- Kochanek C. S. et al., 2012, ApJS, 200, 8
- Kormendy J., Ho L. C., 2013, ARA&A, 51, 511

Koss M. et al., 2017, ApJ, 850, 74 Krogager J. K. et al., 2023, The Messenger, 190, 38 La Marca A. et al., 2024, A&A, 690, A326 Lacy M. et al., 2013, ApJS, 208, 24 Lacy M., Petric A. O., Sajina A., Canalizo G., Storrie-Lombardi L. J., Armus L., Fadda D., Marleau F. R., 2007, AJ, 133, 186 Laigle C. et al., 2016, ApJS, 224, 24 Laloux B. et al., 2023, MNRAS, 518, 2546 Laloux B. et al., 2024, MNRAS, 532, 3459 LaMassa S. M. et al., 2015, ApJ, 800, 144 Lang D., Hogg D. W., Schlegel D. J., 2016, AJ, 151, 36 Lansbury G. B. et al., 2017, ApJ, 846, 20 Laurent P. et al., 2017, J. Cosmol. Astropart. Phys., 2017, 017 Li Z.-J., Dai Y. S., Huang J. S., Wuyts S., Cao T.-W., 2024, ApJ, 963, 99 Liu T. et al., 2022, A&A, 661, A5 Lusso E., Risaliti G., 2016, ApJ, 819, 154 Madau P., Dickinson M., 2014, ARA&A, 52, 415 Magorrian J. et al., 1998, AJ, 115, 2285 Mainieri V. et al., 2005, A&A, 437, 805 Mainzer A. et al., 2014, ApJ, 792, 30 Marconi A., Hunt L. K., 2003, ApJ, 589, L21 Martinez-Sansigre A., Rawlings S., Lacy M., Fadda D., Jarvis M. J., Marleau F. R., Simpson C., Willott C. J., 2006, MNRAS, 370, 1479 Masini A. et al., 2020, ApJS, 251, 2 Mateos S. et al., 2012, MNRAS, 426, 3271 McLure R. J., Jarvis M. J., 2002, MNRAS, 337, 109 Meisner A. M., Lang D., Schlafly E. F., Schlegel D. J., 2019, PASP, 131, 124504 Merloni A. et al., 2012, preprint (arXiv:1209.3114) Merloni A. et al., 2014, MNRAS, 437, 3550 Merloni A. et al., 2019, The Messenger, 175, 42 Merloni A. et al., 2024, A&A, 682, A34 Mullaney J. R., Alexander D. M., Goulding A. D., Hickox R. C., 2011, MNRAS, 414, 1082 Nagao T., Marconi A., Maiolino R., 2006, A&A, 447, 157 Ni Q. et al., 2021, ApJS, 256, 21 Noeske K. G. et al., 2007, ApJ, 660, L43 Oh K. et al., 2018, ApJS, 235, 4 Oliver S. J. et al., 2012, MNRAS, 424, 1614 Pei Y. C., 1992, ApJ, 395, 130

- Petley J. W. et al., 2022, MNRAS, 515, 5159
- Petter G. C. et al., 2022, ApJ, 927, 16
- Petter G. C., Hickox R. C., Alexander D. M., Myers A. D., Geach J. E., Whalen K. E., Andonie C. P., 2023, ApJ, 946, 27
- Powell M. C. et al., 2018, ApJ, 858, 110
- Powell M. C., Urry C. M., Cappelluti N., Johnson J. T., LaMassa S. M., Ananna T. T., Kollmann K. E., 2020, ApJ, 891, 41
- Predehl P. et al., 2021, A&A, 647, A1
- Predehl P., Schmitt J. H. M. M., 1995, A&A, 293, 889
- Prevot M. L., Lequeux J., Maurice E., Prevot L., Rocca-Volmerange B., 1984, A&A, 132, 389
- Rakshit S., Stalin C. S., Kotilainen J., 2020, ApJS, 249, 17
- Ranalli P., Comastri A., Setti G., 2003, A&A, 399, 39
- Rankine A. L., Hewett P. C., Banerji M., Richards G. T., 2020, MNRAS, 492, 4553
- Ricci C., Ueda Y., Koss M. J., Trakhtenbrot B., Bauer F. E., Gandhi P., 2015, ApJ, 815, L13
- Richards G. T. et al., 2006, ApJS, 166, 470
- Rigby J. R. et al., 2018, ApJ, 853, 87
- Rosario D. J., 2019, Astrophysics Source Code Library, record ascl:1904.011
- Ross N. P. et al., 2015, MNRAS, 453, 3932
- Salpeter E. E., 1955, ApJ, 121, 161
- Salvato M. et al., 2022, A&A, 661, A3
- Sanders D. B., Soifer B. T., Elias J. H., Madore B. F., Matthews K., Neugebauer G., Scoville N. Z., 1988, ApJ, 325, 74
- Satyapal S., Secrest N. J., McAlpine W., Ellison S. L., Fischer J., Rosenberg J. L., 2014, ApJ, 784, 113
- Sawicki M., 2012, PASP, 124, 1208

- Schlafly E. F., Meisner A. M., Green G. M., 2019, ApJS, 240, 30 Schnorr-Müller A. et al., 2016, MNRAS, 462, 3570 Scholtz J. et al., 2021, MNRAS, 505, 5469 Schreiber C. et al., 2015, A&A, 575, A74 Shen Y. et al., 2011, ApJS, 194, 45 Shirley R. et al., 2019, MNRAS, 490, 634 Shirley R. et al., 2021, MNRAS, 507, 129 Stern D. et al., 2005, ApJ, 631, 163 Stern D. et al., 2012, ApJ, 753, 30 Stern D. et al., 2014, ApJ, 794, 102 Storey-Fisher K., Hogg D. W., Rix H.-W., Eilers A.-C., Fabbian G., Blanton M. R., Alonso D., 2024, ApJ, 964, 69 Temple M. J. et al., 2023, MNRAS, 523, 646 Temple M. J., Banerji M., Hewett P. C., Rankine A. L., Richards G. T., 2021, MNRAS, 501, 3061 Toba Y. et al., 2022, A&A, 661, A15 Treister E. et al., 2004, ApJ, 616, 123 Trump J. R. et al., 2006, ApJS, 165, 1 Ueda Y., Akiyama M., Hasinger G., Miyaji T., Watson M. G., 2014, ApJ, 786.104 Vestergaard M., Peterson B. M., 2006, ApJ, 641, 689 Wang Ben, Hennawi Joseph F, Cai Zheng, Richards Gordon T, Schindler Jan-Torge, Zakamska Nadia L, Ishikawa Yuzo, Akins Hollis B, Sun Zechang, 2025. Monthly Notices of the Royal Astronomical Society, Werner M. W. et al., 2004, ApJS, 154, 1
- Whitaker K. E., van Dokkum P. G., Brammer G., Franx M., 2012, ApJ, 754, L29
- Wright E. L. et al., 2010, AJ, 140, 1868
- Yan R. et al., 2011, ApJ, 728, 38
- Yan W. et al., 2023, ApJ, 951, 27
- York D. G. et al., 2000, AJ, 120, 1579
- Zakamska N. L. et al., 2003, AJ, 126, 2125
- Zakamska N. L., Gómez L., Strauss M. A., Krolik J. H., 2008, AJ, 136, 1607
- Zakamska N. L., Strauss M. A., Heckman T. M., Ivezić Ž., Krolik J. H., 2004, AJ, 128, 1002
- Zhu G. B. et al., 2015, ApJ, 815, 48

APPENDIX A: *WISE* COLOUR SELECTION PERFORMANCE FOR DIFFERENT *WISE* CATALOGUES

This appendix extends the analysis of Section 3.3 and explores the efficiency of the A18 R90 selection for the AllWISE, unWISE, and LS DR10 *WISE* catalogues. In addition, here we also investigate the origin and nature of the contamination of the AGN candidates. Appendix A1 provides a more detailed description of each *WISE* catalogue, and Appendix A2 presents the properties of the IR AGN candidates selected by A18 R90.

A1 WISE catalogues description

This section describes the data sets used to construct the three *WISE* catalogues used in this work. The AllWISE catalogue contains photometric information for ~ 700 million sources with 5σ point source limits at W1 = 16.83 and W2 = 15.6 mag in the Vega system, identified in the first 6 months of *WISE* observations. The aperture photometry in AllWISE is optimized for point-source characterization, underestimating the flux in extended sources relative to the *WISE* PSF. The unWISE catalogue uses the deep 5-yr image unblurred adds (Meisner et al. 2019), which combines five years of *WISE* and *NEOWISE* (Mainzer et al. 2014) data, and adopts a more careful treatment to model the *WISE* photometry for blended sources, detecting roughly three times as many sources (~ 2 billion sources) as in AllWISE. Lastly, the LS DR10 catalogue uses the position of the *griz*- detected sources to perform forced deblended photometry



Figure A1. IR AGN candidates identified with the Assef et al. (2018) R90 reliability criterion across the W1 - W2 versus W2 plane for $r_{AB} < 22.8$ mag, coloured by the logarithmic 8 – 1000 μ m AGN luminosity obtained from the SED fitting. Grey points are sources without an AGN component in their IR SEDs. We show the selection using the *WISE* photometry from the AllWISE (panel a), unWISE (panel b), and LS DR10 (panel c) catalogues. Each panel reports the AGN reliability of the Assef et al. (2018) selection based on our IR SED fitting.



Figure A2. AGN reliability as a function of the r - W2 colour for the AGN candidates selected by the Assef et al. (2018) R90 AGN colour selection as in Fig. 6, but for three *WISE* catalogues: unWISE (black circle), AllWISE (red star), and LS DR10 (blue square). The left and right panels show the AGN reliabilities for sources with $r_{AB} < 22.1$ mag and < 22.8 mag, respectively. The upper panels show the number of sources per r - W2 bin for the unWISE (black bar), AllWISE (red bar), and LS DR10 (blue bar) catalogues.

in the unWISE co-adds using the *The Tractor* algorithm (Lang, Hogg & Schlegel 2016), providing even deeper *WISE* photometry, reaching 5σ limiting magnitudes at W1 = 20 and W2 = 19.3 mag for ~ 2.8 billion sources.

A2 AGN reliability and contamination of our IR AGN selection

This section extends the analysis presented in Section 3.3 to two additional *WISE* catalogues: AllWISE and LS DR10. Likewise, we use UV-to-FIR SED fitting in four deep fields (see Section 2.3.1) to calculate the AGN reliability and the AGN and host galaxy properties of the A18 R90 AGN candidates identified using the three *WISE*

catalogues. To identify the IR AGN candidates, we follow the A18 R90 AGN wedge (see equation 1). Fig. A1 shows IR AGN candidates identified using the A18 R90 criteria across the W1 - W2 versus W2 plane, and highlights the sources that are also identified as IR AGN by our SED fitting. We find that the A18 selection only reaches 89 per cent reliability when using the AllWISE photometry, and decreases to 66 per cent when using unWISE photometry, and to 52 per cent when using LS DR10 photometry. It is clear from the figure that the LS DR10 catalogue contains more sources with high S/N, and faint W1 and W2 photometry than AllWISE and unWISE. These faint, red sources meet the A18 R90 selection, but are not identified as AGN by our UV-to-FIR SED fitting; i.e. they are contaminant galaxies.

Table A1. AGN reliability and AGN number density (per deg²) of the AGN candidates identified by the A18 R90 selection using the AllWISE, unWISE, and LS DR10 *WISE* photometry (top rows). The AGN reliability is calculated based on the UV-to-FIR SED-fitting results. We also report the values for the obscured Assef et al. (2018) AGN with $r - W2 \ge 5.9$ mag colours (bottom rows).

			AllWISE	unWISE	LS10
A18 R90 AGN	$r_{\rm AB} < 22.8$	AGN candidates density	97	140	219
		AGN reliability	89 per cent	68 per cent	56 per cent
	$r_{\rm AB} < 22.1$	AGN candidates density	84	125	165
		AGN reliability	91 per cent	69 per cent	62 per cent
A18 R90 AGN $+ r - W2 \ge 5.9$	$r_{\rm AB} < 22.8$	AGN candidates density	29	32	73
		AGN reliability	87 per cent	80 per cent	48 per cent
	$r_{\rm AB} < 22.1$	AGN candidates density	17	18	31
		AGN reliability	92 per cent	87 per cent	67 per cent



Figure A3. r - W2 against redshift for the IR AGN candidates identified with the Assef et al. (2018) R90 criterion for $r_{AB} < 22.8$ mag, using AllWISE (panel a), unWISE (panel b), and LS DR10 (panel c) photometry. Grey points are sources without an AGN component in their IR SEDs, and small empty green circles are sources classified as optically extended in the LS DR10 catalogue. The figures show that the majority of the non-SED AGN also have an extended optical morphology.

Fig. A2 further explores the AGN reliability as a function of r - W2, similar to Fig. 6, but this time for the AllWISE, unWISE, and LS DR10 WISE catalogues,²⁴ for $r_{AB} < 22.1$ and $r_{AB} < 22.8$. Fig. A2 shows that when using AllWISE photometry, the AGN reliability does not have a strong dependence on r - W2. However, when using the unWISE photometry, the A18 R90 colour-selection approach provides a highly contaminated sample at r - W2 < 4 mag colours, with an AGN reliability of ≈ 20 per cent, but it drastically improves to > 80 per cent for $r - W2 \gtrsim 4.5$ mag. At $r - W2 \geq$ 5.9 mag, which is our obscured AGN selection, the A18 R90 selection provides a clean AGN sample with an AGN reliability of 87 per cent for $r_{AB} < 22.1$ and 80 per cent for $r_{AB} < 22.8$. Finally, when using the WISE photometry from the LS DR10, the AGN reliability is < 60 per cent for most r - W2 colour bins, including r - W2 > 5.9 mag, indicating that the A18 AGN selection does not perform well when using LS WISE photometry.

Table A1 provides a summary of the AGN reliability and number density of the IR AGN candidates identified by A18 R90 for different *WISE* catalogues. We note that the AGN density is not always the

same: when using the *WISE* magnitudes from the unWISE and LS10 catalogues, the A18 R90 selection identifies $\sim 1.4 \times$ and $\sim 2.3 \times$ more AGN than when using the AllWISE catalogue, respectively. However, we find that most of the additional IR AGN candidates are galaxies misidentified as IR AGN. After the $r - W2 \ge 5.9$ mag colour cut is applied, unWISE finds only a slightly higher AGN number density than AllWISE, probably due to the fact the colour cut removed most of the non-AGN contamination from the unWISE sample.

Fig. A3 plots the A18 R90 IR AGN candidates across the r - W2versus redshift plane, highlighting the sources that are identified as AGN by our SED fitting and also the sources that have an extended optical morphology in the LS DR10. We find that for the three catalogues, the contaminant sources are typically morphologically classified as optically extended in the LS DR10 catalogue. When using WISE photometry from the AllWISE catalogue, \approx 76 per cent of the non-AGN contaminants are optically extended. When using WISE photometry from the unWISE and LS DR10 catalogues, > 90 per cent of the non-AGN contaminants are optically extended, and in consequence, the AGN reliability of the A18 R90 significantly drops to $\approx 30 - 40$ per cent. However, for optically pointlike sources, the AGN reliability for the unWISE and LS DR10 catalogues is about 90 per cent, as shown by Assef et al. (2018). Table A2 summarizes the AGN reliability for each aforementioned subsample.

Fig. A4 examines the host galaxy properties of the non-AGN contaminants. For the three catalogues, the majority (> 50 per cent)

²⁴We note that A18 used UV-to-MIR SED fitting to arrive at their 90 per cent AGN reliability threshold using AllWISE photometry (see Assef et al. 2013, for details on their SED-fitting approach). In our work, we also independently assess the reliability of the approach using other deeper *WISE* photometry, such as unWISE and the LSDR10, and we use a more sophisticated SED modelling which also includes the FIR band.



Figure A4. Redshift (left panel), stellar masses (central panel), and SFRs (right panel) of the non-AGN contaminants identified from our UV-to-FIR SED fitting of the IR AGN candidates selected with the Assef et al. (2018) R90 criterion using AllWISE (dashed blue), unWISE (solid green), and LS DR10 (dotted red) catalogues.

Table A2. AGN reliability of the A18 R90 selection for sources with z < 0.5, sources that are classified as optically extended by the LS DR10 catalogue, and sources that are classified as optically point-like by the LS DR10 catalogue. We show the results for the three different *WISE* catalogues explored in this work.

A18 R90 AGN subsample	AllWISE (per cent)	unWISE (per cent)	LS10 (per cent)
Entire sample	89	68	52
z < 0.5	79	35	30
Extended optical morphology	79	41	31
Point-like source	96	96	88

of the sources have z < 0.5. The stellar masses of the non-AGN contaminants are not so different from the SED AGN, ranging $\approx 10^{10} - 10^{12} \text{ M}_{\odot}$. However, the non-AGN contaminants are systems with little ongoing SF, with more than 80 per cent of the sources having SFR $< 10 \text{ M}_{\odot} \text{ yr}^{-1}$. These results indicate that the non-AGN contaminants are typically low redshift, quiescent galaxies, with red optical colours.

In order to understand the higher number of 'apparent' AGN in the unWISE and LS DR10 catalogues, we visually inspect a random sample of optically bright, extended IR AGN selected by A18 R90. We find that the unWISE and LS DR10 catalogues split the extended sources into one central source plus several fainter components. This is a result of the deblending approach used to extract the photometry, which is acknowledged in Schlafly et al. (2019). Hence, a significant part of the AGN contamination for the unWISE and LS DR10 catalogues is due to the misidentification of faint and red extended components as AGN. A solution to increase the purity of the A18 R90 AGN samples identified using unWISE and LS DR10 photometry would be only to apply the A18 R90 selection to point-like sources in the LS DR10 catalogue. However, that would significantly decrease the completeness of the AGN samples as only \sim 46 per cent and \sim 38 per cent of the A18 R90 AGN have point-like morphologies in the unWISE and LS DR10 catalogues, respectively.

We also investigate the overlap of the different *WISE* catalogues. Fig. A5 shows Venn diagrams of the A18 R90 selected AGN using the AllWISE, unWISE, and LS DR10 catalogues for all the sources, the sources that are also identified as AGN by the UV-to-FIR SED fitting, and the sources with $r - W2 \ge 5.9$ mag. We find that ~ 78 per cent of the AGN selected using the AllWISE photometry are also selected when using the unWISE and LS DR10 photometry. The overlap increases to 86 per cent when we only consider the sources that are also identified as AGN by UV-to-FIR SED fitting. The numbers are the same when considering only the sources with $r - W2 \ge 5.9$ mag colours. This analysis shows that the unWISE and the LS DR10 *WISE* photometry find more IR AGN candidates than AllWISE, but the three catalogues find almost the same real AGN.

Overall, since we are interested in the obscured IR AGN candidates with r - W2 > 5.9 mag, we conclude that the unWISE and AllWISE photometry are both good options for selecting IR AGN when using the A18 R90 approach, as both provide high AGN reliabilities (i.e. > 80 - 90 per cent; see Fig. A2 and Table A1) and similar AGN number densities. The LS DR10 also provides a clean AGN sample when only considering the sources with point-like optical morphologies but at the expense of significantly reducing the completeness of the AGN sample. Based on these results, we decided to use the unWISE catalogue to perform the target selection of the 4MOST IR AGN survey for three reasons: (1) it provides a clean AGN sample, finding AGN with 80 per cent and 87 per cent reliability down to $r_{AB} = 22.8$ and 22.1, respectively; (2) its slightly higher AGN number density (it finds 1-2 more real AGN per deg²) provides a more complete AGN sample, adding $\sim 11\,000$ sources in the final survey, and (3) it is more closely aligned with the LS DR10 catalogue, which is our parent sample, since it uses the same unWISE image co-adds and a consistent WISE photometric extraction approach (see Appendix A1 for details). Hence, the un-WISE photometry provides the best balance between reliability and completeness.



Figure A5. Venn diagrams showing the overlap of A18 R90 AGN candidates (labelled 'Everything') using AllWISE, unWISE, and LS DR10 photometry (left plots) and the same overlap only for the sources that are also classified as IR AGN by our UV-to-FIR SED fitting (labelled 'SED AGN', right plots). The bottom plots show the overlap only for sources with $r - [4.5] \ge 5.9$ mag colours. In the four cases, the majority (~ 78 per cent) of the AGN identified using the AllWISE photometry are also identified when using the unWISE and LS DR10 photometry.

APPENDIX B: SED FITTING OF THE 4MOST IR AGN SURVEY SOURCES WITH UV-TO-MIR PHOTOMETRY

In this paper, we have used SED fitting in well-observed fields in the sky with X-ray and UV-to-FIR photometry to predict the AGN and host galaxy properties of the 4MOST IR AGN survey sample. However, for the majority of the sources in the 4MOST IR AGN survey, we will be restricted to just 6–8 photometric measurements (i.e. the LS bands *griz*, and the *WISE* bands *W*1, *W*2, *W*3, and *W*4) in the UV-to-MIR band to perform SED fitting. This all-sky available photometry is limited, but, as we demonstrate below in this section, it can still reliably constrain the AGN torus emission and the stellar mass for obscured AGN.

To fit the all-sky photometry, we use a more simplified SED-fitting model to the one described in Section 2.3.2. This time, we reduce the degree of free parameters for the stellar population and torus models. For the stellar population emission, we consider the simple stellar population model from Bruzual & Charlot (2003), which assumes

one single instantaneous starburst at $\tau = 0$. For the torus emission, we only allow the short-wavelength slope to be free to vary while fixing the long-wavelength slope to $\Gamma = -0.9$, the best-fitting value found in our more-detailed SED template fitting in the deep fields (see Section 2.3).

To help guide the simplified SED-fitting modelling, we use the knowledge gained from the AGN and host galaxy property constraints from the deep fields. For this, we take advantage of the Bayesian nature of FORTESFIT, and place priors on the AGN luminosity and stellar mass fitting parameters. To calculate the priors, we use the deep-field results and fit Gaussian functions to the distributions of both parameters for different redshift ranges; see Table B1 for the best-fitting parameters (i.e. mean μ and standard deviation σ) of the AGN luminosity and stellar masses. We note that the standard deviation is typically $\sigma \approx 0.4$ dex, which gives the SEDfitting process freedom to explore a wide range of values around the central position of the Gaussian.

We tested this approach in the XMM-LSS and ELAIS-S1 fields. We excluded the Boötes field from the analysis because it does



Downloaded from https://academic.oup.com/mnras/article/539/3/2202/8115791 by guest on 20 May 2025

Figure B1. Example best-fitting SED models for an X-ray-detected (top panels) and an X-ray-undetected (bottom panels) obscured IR AGN candidates in the XMM-LSS field. The left figures show the best-fitting SED using all the available UV-to-FIR photometry, and the left figures show the best-fitting SED only using LSDR10 and *WISE* photometry. Black points with error bars are the photometric data with their 1 σ uncertainties, and red empty circles are the best-fitting model photometry. The blue curves represent the dust-obscured SF component, the red curves are the torus components, the orange curves are the stellar population emission, and the green curves are the AD emission. The spread in the curves represents the approximate 1 σ scatter in the SED components as constrained by FORTESFIT. The black line is the sum of all the best-fitting components. The magenta outline box shows the best-fitting AGN luminosity and stellar masses and their 1 σ uncertainties. The ID and redshifts are as in the HELP catalogue (Shirley et al. 2019, 2021). These figures show samples of the full SED and are only meant to guide the reader; hence, they should not be used to measure the merit of the fit.

Table B1. Gaussian priors placed on the simple LS DR10 plus *WISE* SED fitting. The values reported in this table correspond to the best-fitting mean (μ) and standard deviation (σ) for the predicted AGN luminosity, stellar masses, and SF luminosities for different redshifts bins of the 4MOST IR AGN sources, calculated from the detailed UV-to-FIR SED fitting in the deep fields (see Sections 2.3.2and 4.2).

Parameter	Redshift	Gaussian prior parameters
$\log L_{ m AGN, IR}$	z < 1 $z = 1 - 2$ $z > 2$	$\mu = 44.9 \text{ erg s}^{-1}, \sigma = 0.41$ $\mu = 45.5 \text{ erg s}^{-1}, \sigma = 0.44$ $\mu = 46.5 \text{ erg s}^{-1}, \sigma = 0.42$
$\log M_{\star}$	z < 0.5 z = 0.5 - 1 z = 1 - 1.5 z = 1.5 - 2 z = > 2	$\begin{split} \mu &= 10.2 \ \mathrm{M}_{\odot}, \ \sigma = 0.5 \\ \mu &= 10.8 \ \mathrm{M}_{\odot}, \ \sigma = 0.4 \\ \mu &= 11.0 \ \mathrm{M}_{\odot}, \ \sigma = 0.4 \\ \mu &= 11.3 \ \mathrm{M}_{\odot}, \ \sigma = 0.4 \\ \mu &= 11.9 \ \mathrm{M}_{\odot}, \ \sigma = 0.9 \end{split}$
$\log L_{\rm SF,IR}$	z < 0.5 z = 0.5 - 1 z = 1 - 1.5 z > 2	$\mu = 45.0 \text{ erg s}^{-1}, \sigma = 0.5$ $\mu = 45.3 \text{ erg s}^{-1}, \sigma = 0.5$ $\mu = 45.9 \text{ erg s}^{-1}, \sigma = 0.6$ $\mu = 46.0 \text{ erg s}^{-1}, \sigma = 0.4$

not have *i*-band photometry, which is not the case for the 4MOST IR AGN survey sources. Fig. B1 shows two examples of bestfitting SED models for an X-ray-undetected (top) and detected (bottom) obscured IR AGN candidates in the XMM-LSS field. The figure shows the best SED-fitting results when using all the available UV-to-FIR photometry (left) and only the LSDR10 and WISE photometry. The AGN luminosity and the stellar masses are consistent within 1σ between the two SED-fitting implementations. We note that with the current UV-to-MIR available photometry, it is not possible to constrain the contribution from the AD and dustobscured SF. Both components are well constrained in the UV-to-MIR SED modelling only because we set priors in the AD and SF luminosities, which the SED fitting is assuming as truthful given the lack of photometry in the UV and FIR wavelengths. When available, we can use an X-ray luminosity prior to constrain the AD luminosity. However, constraining the dust-obscured SFRs will be challenging as the majority of the IR AGN wide survey footprint does not have FIR photometric coverage. Fig. B2 shows the comparison of the AGN luminosity and stellar masses when using only LSDR10 and WISE photometry (y-axis), and when using



Figure B2. Comparison of SED-fitting constraints when only using the UV-to-MIR LS+*WISE* photometry (*y*-axis), and when using all the available UV-to-FIR photometry (*x*-axis). The left panel compares the 8 – 1000 μ m AGN luminosity ($L_{AGN,IR}$) and the right panel compares the stellar mass (M_{\star}). Red circles are sources with z < 0.5, green circles sources with z = [0.5, 1], blue circles sources with z = [1, 1.5], orange circles sources with z = [1.5, 2], and black circles sources with z > 2. The small panel on the right-bottom corner of each plot shows the distribution of the difference between the parameter values obtained using UV-to-FIR photometry and only UV-to-MIR photometry.

all the available UV-to-FIR photometry (*x*-axis). Overall, we find a good agreement between both SED-fitting values; however, we still find many sources where both values are inconsistent. Hence, this approach will be more useful to constrain the average properties of a given sample. We note that when having additional AD or stellar masses constraints provided by the 4MOST optical spectra

and/or X-ray constraints, we will be able to use that information to improve even more the SED-fitting priors, producing more accurate results.

This paper has been typeset from a TEX/LATEX file prepared by the author.